Arthur's multiplicity formula for certain inner forms of special orthogonal and symplectic groups

Olivier Taïbi *

Abstract

Let **G** be a special orthogonal group or an inner form of a symplectic group over a number field F such that there exists a non-empty set S of real places of F at which **G** has discrete series and outside of which **G** is quasi-split. We prove Arthur's multiplicity formula for automorphic representations of **G** having algebraic regular infinitesimal character at all places in S.

Contents

1	Intr	roduction	2	
2	Rev	view of Arthur's results	4	
	2.1	Quasi-split symplectic and special orthogonal groups and their Langlands		
		dual groups	4	
	2.2	Self-dual automorphic cuspidal representations of \mathbf{GL}_N	5	
	2.3	Elliptic endoscopic data and embeddings of L-groups	5	
	2.4	Arthur's substitute for global parameters	7	
	2.5	Local Arthur packets in the quasi-split case	8	
	2.6	Localisation of global parameters	10	
3	Certain rigid inner forms and Arthur packets		10	
	3.1	Certain rigid inner forms	10	
		3.1.1 Rigid inner forms at real places	11	
		3.1.2 Global rigid inner forms	14	
	3.2	Local Arthur packets for real reductive groups having discrete series	17	
		3.2.1 Tempered parameters	17	
		3.2.2 Non-tempered parameters: Adams-Johnson packets	20	
		3.2.3 Relation with Arthur packets	22	
	3.3	Trivial inner twists	23	
	3.4	Adèlic Arthur packets for rigid inner forms	25	
4	Art	hur's multiplicity formula for certain inner forms	27	

^{*}Imperial College London, United Kingdom. The author is supported by ERC Starting Grant 306326.

1 Introduction

Let F be a number field, and consider **G** a special orthogonal group or an inner form of a symplectic group over F. Assume that there exists a *non-empty* set S of real places of F such that

- for any place $v \in S$, the group $\mathbf{G}(F_v)$ admits discrete series representations,
- for any place $v \notin S$, the reductive group \mathbf{G}_{F_v} is quasi-split.

This paper is devoted to the proof of Arthur's multiplicity formula for the subspace

$$L^2_{\text{disc}}(\mathbf{G}(F) \setminus \mathbf{G}(\mathbb{A}_F))^{S-\text{alg. reg.}}$$

of the discrete automorphic spectrum $L^2_{\text{disc}}(\mathbf{G}(F)\backslash\mathbf{G}(\mathbb{A}_F)) \subset L^2(\mathbf{G}(F)\backslash\mathbf{G}(\mathbb{A}_F))$ corresponding to algebraic regular representations ("C-algebraic regular" in the sense of [BG10]) at all places in S, that is Theorem 4.0.1, using the stabilisation of the trace formula for \mathbf{G} . As in the case of quasi-split groups [Art13, Theorem 1.5.2], in the even orthogonal case we obtain a multiplicity formula only "up to outer automorphism at all places of F".

In particular we obtain the multiplicity formula for the full automorphic spectrum under the stronger assumption that $\mathbf{G}(\mathbb{R} \otimes_{\mathbb{Q}} F)$ is compact and G is quasi-split at all finite places F. The interest of this case is that automorphic forms for such a group \mathbf{G} are more concrete than for quasi-split groups. This allows one to compute explicitly and relatively easily with spaces of such automorphic forms. Moreover, this assumption on $\mathbf{G}(\mathbb{R} \otimes_{\mathbb{Q}} F)$ allows for a much simpler theory of p-adic families of automorphic forms: see [Che04], [Loe11]. Several papers already rely on Theorem 4.0.1 for groups \mathbf{G} satisfying this stronger assumption: [Taïb] (and thus also [CLH]), [CR14] (see Conjectures 3.26 and 3.30), and [CL] (see Conjecture 4.25).

The multiplicity formula for arbitrary inner forms of symplectic and special orthogonal groups over number fields was announced by Arthur in [Art13, Chapter 9], but even formulating the result precisely was not possible at the time because of the absence of canonical absolute transfer factors as in the quasi-split case, which were only recently constructed by Kaletha [Kalb], [Kala]. Furthermore, the general case will require the construction of local Arthur packets for non-quasi-split groups, as well as solving the problem, observed by Arthur, that some inner forms of even special orthogonal groups do not admit an outer automorphism defined over the base field. In contrast, the above assumptions on **G** allow us to give a simple proof of the multiplicity formula, thanks to the following simplifications.

- If **G** is an even special orthogonal group, i.e. a *pure* inner form of a quasi-split group rather than an arbitrary inner form, then the non-trivial outer automorphism of **G** has a representative defined over F. In fact there is a natural $\mathbf{G}(F)$ -orbit of such automorphisms.
- At the finite places of F, we can use Arthur packets for quasi-split groups constructed in [Art13].

The assumption on G at places in S and the "algebraic regular" assumption imply that the only relevant Arthur-Langlands parameters at these places are those considered by Adams and Johnson in [AJ87], that is parameters bounded on the Weil group of ℝ and having algebraic regular infinitesimal character. This simplifies significantly the spectral side in the stabilisation of the trace formula, since we do not have to consider contributions from non-discrete parameters to the discrete part of the spectral side of the stabilisation of the trace formula as in the general case.

For applications, using arbitrary inner forms instead of quasi-split groups is constraining for at least two reasons.

The first reason is that at the non-quasi-split places it is in general difficult to describe precisely the local Arthur packets, or even to establish whether a given Arthur packet is non-empty, contrary to the quasi-split case where Arthur packets contain the corresponding Langlands packets by [Art13, Proposition 7.4.1]. This is not an issue in our case, as we will make the internal parametrisation of Adams-Johnson packets explicit using Shelstad's parametrisation of tempered L-packets ([She08a], [She10], [She08b]) and Kaletha's refinement of Shelstad's formulation in [Kalb, §5.6]. We will be able to appeal to [AJ87] thanks to recent work of Arancibia, Moeglin and Renard [AMR] which proves that for Adams-Johnson parameters and quasi-split symplectic or special orthogonal groups, Arthur packets coincide with Adams-Johnson packets. In particular these packets are non-empty.

The second reason that makes arbitrary inner forms less practical is that contrary to the quasi-split case it is not true that any *tempered* Arthur-Langlands parameter gives rise to at least one automorphic representation, namely the generic one (with respect to a choice of Whittaker datum). Again the fact that Arthur packets at the real places of F are completely explicit can allow to circumvent this difficulty in situations where solvable base change is possible, as in [Taïb].

Since Mok [Mok15] did for quasi-split unitary groups what Arthur did for quasi-split symplectic and special orthogonal groups, and Arancibia, Moeglin and Renard also treated the case of unitary groups in [AMR], the same method as in the present paper could be used to prove the analogue of Theorem 4.0.1 for inner forms of unitary groups for which there exists a non-empty set S of real places of F satisfying the above assumption. In fact this case would be easier since there are no issues with outer automorphisms as in the even orthogonal case. However, for *pure* inner forms of unitary groups and *tempered* parameters Kaletha, Minguez, Shin and White have already proved Arthur's multiplicity formula in [KMSW], and they also announced a proof of the general case. They used a slightly different formulation, using Kaletha's extended pure inner twists instead of Kaletha's rigid inner twists that we use in the present paper. We refer the interested reader to [Kalc] for a comparison of these notions.

Let us end this introduction with notations that we will use throughout the paper. For K a perfect field we will denote by Gal_K the absolute Galois group $\operatorname{Gal}(\overline{K}/K)$ of K. If K is a local or global field of characteristic zero, W_K will denote its Weil group. If K is a local field of characteristic zero, WD_K will denote its Weil-Deligne group: WD_K = W_K if

 $K \simeq \mathbb{R}$ or \mathbb{C} , and $WD_K = W_K \times SU(2)$ if K is p-adic.

2 Review of Arthur's results

In this section we briefly review Arthur's results from [Art13], i.e. the construction of local and adélic Arthur packets for *quasi-split* special orthogonal and symplectic groups.

2.1 Quasi-split symplectic and special orthogonal groups and their Langlands dual groups

Let F be a global or local field of characteristic zero. As in [Art13], we will consider three families of quasi-split reductive groups over F:

- for $n \ge 0$, the split symplectic group \mathbf{Sp}_{2n} , defined as the stabilizer of a nondegenerate alternating form on a vector space of dimension 2n over F,
- for $n \ge 1$, the split special orthogonal group \mathbf{SO}_{2n+1} , defined as follows. Let V be a vector space of dimension 2n+1 over F, and let q be a maximally split non-degenerate quadratic form q on V: (V,q) is the orthogonal direct sum of n hyperbolic planes and a line. Then \mathbf{SO}_{2n+1} is the stabilizer of q in the special linear group $\mathbf{SL}(V)$.
- for n ≥ 1 and α ∈ F[×]/F^{×2}, the quasi-split special orthogonal group SO^α_{2n}, defined as follows. Let E = F[X]/(X² − α) and consider the quadratic form q_α : x → N_{E/F}(x) on E seen as a two-dimensional vector space over F. Note that (E, q_α) is a hyperbolic plane if and only if α = 1, i.e. α is a square. Let (V, q) be the orthogonal direct sum of n − 1 hyperbolic planes and (E, q_α). Then SO^α_{2n} is the stabilizer of q in the special linear group SL(V).

In the third case we will also see α as a continuous character $\mathbb{A}_F^{\times}/F^{\times} \to \{\pm 1\}$ if F is global $(\alpha : F^{\times} \to \{\pm 1\})$ if F is local, using class field theory.

Denote by **G** one of the above groups. As in [Bor79] denote by ${}^{L}\mathbf{G}$ the Langlands dual group of **G** and $\widehat{\mathbf{G}}$ its identity component, so that

$$\widehat{\mathbf{G}} = \begin{cases} \mathrm{SO}_{2n+1}(\mathbb{C}) & \text{if } \mathbf{G} = \mathbf{Sp}_{2n} \\ \mathrm{Sp}_{2n}(\mathbb{C}) & \text{if } \mathbf{G} = \mathbf{SO}_{2n+1} \\ \mathrm{SO}_{2n}(\mathbb{C}) & \text{if } \mathbf{G} = \mathbf{SO}_{2n}^{\alpha} \end{cases}$$

and ${}^{L}\mathbf{G} = \widehat{\mathbf{G}} \rtimes \operatorname{Gal}_{F}$. The action of Gal_{F} on $\widehat{\mathbf{G}}$ is trivial except when $\mathbf{G} = \mathbf{SO}_{2n}^{\alpha}$ with α nontrivial, in which case the action of Gal_{F} factors through $\operatorname{Gal}(E/F) = \{1, \sigma\}$ and σ acts by outer conjugation on $\widehat{\mathbf{G}}$. To be completely explicit, one can identify $\widehat{\mathbf{G}} \rtimes \operatorname{Gal}(E/F)$ with $\operatorname{O}_{2n}(\mathbb{C})$ as follows. Let B be the non-degenerate symmetric bilinear form on $\mathbb{C}v_1 \oplus \cdots \oplus \mathbb{C}v_{2n}$ defined by $B(v_i, v_j) = \delta_{i,2n+1-j}$, then $1 \rtimes \sigma$ is identified with the element of $\operatorname{O}_{2n}(\mathbb{C})$ fixing $v_1, \ldots, v_{n-1}, v_{n+2}, \ldots, v_{2n}$ and exchanging v_n and v_{n+1} .

Denote by $\operatorname{Std}_{\mathbf{G}}$ the standard representation $\widehat{\mathbf{G}} \to \operatorname{GL}_N(\mathbb{C})$ where

$$N = N(\widehat{\mathbf{G}}) = \begin{cases} 2n & \text{if } \mathbf{G} = \mathbf{SO}_{2n+1} \text{ or } \mathbf{SO}_{2n}^{\alpha}, \\ 2n+1 & \text{if } \mathbf{G} = \mathbf{Sp}_{2n}. \end{cases}$$

Note that we have included the trivial group \mathbf{Sp}_0 in the list above because the standard representation of its dual group has dimension 1. Extend $\operatorname{Std}_{\mathbf{G}}$ to ${}^{L}\mathbf{G}$ as follows.

- If G is symplectic, odd orthogonal, or split even orthogonal (that is α = 1), let Std_G be trivial on Gal_F.
- Otherwise $\mathbf{G} = \mathbf{SO}_{2n}^{\alpha}$ with $\alpha \neq 1$. Let $\operatorname{Std}_{\mathbf{G}}$ be trivial on Gal_{E} . If n = 1, up to conjugation by $\operatorname{Std}_{\mathbf{G}}(\widehat{\mathbf{G}})$ there is a unique choice for $\operatorname{Std}_{\mathbf{G}}(1 \rtimes \sigma)$. If n > 1, there are two possibilities for $\operatorname{Std}_{\mathbf{G}}(1 \rtimes \sigma)$, choose the one having -1 as an eigenvalue with multiplicity one (the other choice has -1 as an eigenvalue with multiplicity N 1 = 2n 1). Via the above identification $\widehat{\mathbf{G}} \rtimes \operatorname{Gal}(E/F) \simeq \operatorname{O}_{2n}(\mathbb{C})$, this choice for $\operatorname{Std}_{\mathbf{G}}$ is simply the natural inclusion $\operatorname{O}_{2n}(\mathbb{C}) \subset \operatorname{GL}_{2n}(\mathbb{C})$.

Finally, let $\operatorname{Aut}({}^{L}\mathbf{G})$ be the group of automorphisms of ${}^{L}\mathbf{G}$, and $\operatorname{Out}({}^{L}\mathbf{G}) = \operatorname{Aut}({}^{L}\mathbf{G})/\widehat{\mathbf{G}}_{ad}$. Note that $\operatorname{Out}({}^{L}\mathbf{G})$ is trivial unless $\mathbf{G} = \mathbf{SO}_{2n}^{\alpha}$, in which case it is isomorphic to $\mathbb{Z}/2\mathbb{Z}$. Even in this case, the $\operatorname{GL}_{N}(\mathbb{C})$ -conjugacy class of $\operatorname{Std}_{\mathbf{G}}$ is invariant under $\operatorname{Out}({}^{L}\mathbf{G})$.

2.2 Self-dual automorphic cuspidal representations of \mathbf{GL}_N

In this section we let F be a number field. Let $N \ge 1$, and consider an automorphic cuspidal representation π of \mathbf{GL}_N/F which is self-dual. Arthur [Art13, Theorem 1.4.1] associates a quasi-split special orthogonal or symplectic group \mathbf{G}_{π} with π , which satisfies $N(\widehat{\mathbf{G}}_{\pi}) = N$. The central character ω_{π} of π has order 1 or 2, and is trivial if $\widehat{\mathbf{G}}_{\pi}$ is symplectic. If N is odd, $\pi \otimes \omega_{\pi}$ is also self-dual and has trivial central character, and $\mathbf{G}_{\pi} = \mathbf{Sp}_{N-1}$. Choose $\mathrm{Std}_{\pi} : {}^{L}\mathbf{G}_{\pi} \to \mathrm{GL}_{N}(\mathbb{C})$ extending the standard representation $\widehat{\mathbf{G}}_{\pi} \to \mathrm{GL}_{N}(\mathbb{C})$:

- If $\widehat{\mathbf{G}}_{\pi}$ is odd orthogonal, Std_{π} is the twist of $\operatorname{Std}_{\mathbf{G}_{\pi}}$ (defined in the previous section) by the character $\operatorname{Gal}_{F} \to \{\pm 1\}$ corresponding to ω_{π} by class field theory.
- Otherwise let $\operatorname{Std}_{\pi} = \operatorname{Std}_{\mathbf{G}_{\pi}}$.

We can now state the remaining part of [Art13, Theorem 1.4.1 and 1.4.2]: for any place v of F, the local Langlands parameter $WD_{F_v} \to GL_N(\mathbb{C})$ of π_v is conjugate to $Std_{\pi} \circ \phi_v$ for some Langlands parameter $\phi_v : WD_{F_v} \to {}^L\mathbf{G}_{\pi}$. Moreover the Aut $({}^L\mathbf{G}_{\pi})$ -orbit of ϕ_v is determined by the Langlands parameter of π_v .

Define $\operatorname{sign}(\widehat{\mathbf{G}}) = +1$ (resp. -1) if $\widehat{\mathbf{G}}$ is orthogonal (resp. symplectic), and define $\operatorname{sign}(\pi) = \operatorname{sign}(\widehat{\mathbf{G}}_{\pi})$.

2.3 Elliptic endoscopic data and embeddings of L-groups

In this section F denotes a local or global field of characteristic zero, and \mathbf{G} denotes one of the groups defined in section 2.1 Let us recall from [Wal10, §1.8] the isomorphism classes of elliptic endoscopic data for \mathbf{G} , and fix an embedding of L-groups in each case. To avoid confusion we ought to be precise about the terminology that we will use, as there is a slight discrepancy between [KS99] and [Wal10]. We will use the definition of an endoscopic datum ($\mathbf{H}, \mathcal{H}, s, \xi$) for \mathbf{G} given in [KS99, §2.1]. In particular, \mathcal{H} is a splittable extension of W_F by $\widehat{\mathbf{H}}$ and $\xi : \mathcal{H} \to {}^{L}\mathbf{G}$ is an L-embedding. Waldspurger also includes an L-embedding ${}^{L}\xi : {}^{L}\mathbf{H} \to {}^{L}\mathbf{G}$ which is the composition of $\xi : \mathcal{H} \to {}^{L}\mathbf{G}$ and an isomorphism ${}^{L}\mathbf{H} \to \mathcal{H}$. As observed in [KS99, §2.2], in general such an isomorphism does not necessarily exist, a problem which is overcome by introducing a z-extension of \mathbf{H} . Fortunately such an isomorphism does exist for any endoscopic datum of any group \mathbf{G} as in section 2.1, and thus we need not consider such z-extensions.

Here is the list of all isomorphism classes of elliptic endoscopic data for the groups defined in section 2.1. For simplicity we only give \mathbf{H} , and refer to [Wal10, §1.8] for more details. We also give the group $\operatorname{Out}(\mathbf{H}, \mathcal{H}, s, \xi) = \operatorname{Aut}(\mathbf{H}, \mathcal{H}, s, \xi)/\hat{\mathbf{H}}$ of outer automorphisms in each case.

- $\mathbf{G} = \mathbf{SO}_{2n+1}$: $\mathbf{H} = \mathbf{SO}_{2a+1} \times \mathbf{SO}_{2b+1}$ with a + b = n. The pairs (a, b) and (b, a) define isomorphic endoscopic data. The outer automorphism group of $(\mathbf{H}, \mathcal{H}, s, \xi)$ is trivial except when a = b, where there is a unique non-trivial outer automorphism, swapping the two factors of $\widehat{\mathbf{H}} = \mathrm{Sp}_{2a}(\mathbb{C}) \times \mathrm{Sp}_{2b}(\mathbb{C})$.
- $\mathbf{G} = \mathbf{Sp}_{2n}$: $\mathbf{H} = \mathbf{Sp}_{2a} \times \mathbf{SO}_{2b}^{\alpha}$ with a + b = n and $(b, \alpha) \neq (1, 1)$. We impose that $\alpha = 1$ if b = 0. There is a non-trivial outer automorphism if and only if b > 0, in which case the unique non-trivial automorphism acts by outer conjugation on the factor $\mathrm{SO}_{2b}(\mathbb{C})$ of $\widehat{\mathbf{H}}$.
- $\mathbf{G} = \mathbf{SO}_{2n}^{\alpha}$: $\mathbf{H} = \mathbf{SO}_{2a}^{\beta} \times \mathbf{SO}_{2b}^{\gamma}$, with a + b = n, $\beta = 1$ if a = 0, $\gamma = 1$ if b = 0, $\beta\gamma = \alpha$, and both (a, β) and (b, γ) are distinct from (1, 1). The pairs $((a, \beta), (b, \gamma))$ and $((b, \gamma), (a, \beta))$ define isomorphic endoscopic data. If ab > 0 there is a nontrivial outer automorphism acting by simultaneous outer conjugation on the two factors of $\widehat{\mathbf{H}}$. There are no other non-trivial outer automorphisms except when a = b and $\alpha = 1$, in which case there is one swapping the two factors of $\widehat{\mathbf{H}}$, and so $\operatorname{Out}(\mathbf{H}, \mathcal{H}, s, \xi) \simeq \mathbb{Z}/2\mathbb{Z} \times \mathbb{Z}/2\mathbb{Z}$ in this case.

As explained above, in each case Waldspurger also chooses a particular L-embedding ${}^{L}\xi$: ${}^{L}\mathbf{H} \rightarrow {}^{L}\mathbf{G}$. This choice is somewhat arbitrary, as it could be twisted by a 1-cocycle $W_{F} \rightarrow Z(\hat{\mathbf{H}})$, and all results requiring the theory of endoscopy (for example the endoscopic character relation (2)) ought to be valid for any choice of embedding. However, as in [Art13] it will be convenient to fix these embeddings, at least in the global setting, and we refer to [Wal10, §1.8] for their definition. It is important to note that Waldspurger's choice of embedding ${}^{L}\xi$, a priori defined only for a particular representative $(\mathbf{H}, \mathcal{H}, s, \xi)$ in its isomorphism class of elliptic endoscopic data, has the benefit of being invariant under $\operatorname{Out}(\mathbf{H}, \mathcal{H}, s, \xi)$. Consequently this gives a well-defined embedding ${}^{L}\xi'$ for any endoscopic datum $(\mathbf{H}', \mathcal{H}', s', \xi')$ isomorphic to $(\mathbf{H}, \mathcal{H}, s, \xi)$.

It is not difficult to check directly on the definition of transfer factors that the validity of statements such as (2) does not depend on the choice of embedding.

Finally, let us determine the relationship between Waldspurger's choice of embeddings and the standard representations defined in 2.1, which are embeddings for twisted endoscopy for general linear groups. Write $\mathbf{H} = \mathbf{H}_1 \times \mathbf{H}_2$. If **G** is orthogonal or $\mathbf{G} = \mathbf{Sp}_{2n}$ and $\mathbf{H} = \mathbf{Sp}_{2a} \times \mathbf{SO}_{2b}^{\alpha}$ with $\alpha = 1$, then $\operatorname{Std}_{\mathbf{G}} \circ {}^{L}\xi$ is conjugated to $(\operatorname{Std}_{\mathbf{H}_{1}} \oplus \operatorname{Std}_{\mathbf{H}_{2}}) \circ \iota$, where ι denotes the inclusion ${}^{L}\mathbf{H} \subset {}^{L}\mathbf{H}_{1} \times {}^{L}\mathbf{H}_{2}$. In the remaining case where $\mathbf{G} = \mathbf{Sp}_{2n}$ and $\mathbf{H} = \mathbf{Sp}_{2a} \times \mathbf{SO}_{2b}^{\alpha}$ with $\alpha \neq 1$, $\operatorname{Std}_{\mathbf{G}} \circ {}^{L}\xi$ is conjugated to $((\alpha \otimes \operatorname{Std}_{\mathbf{Sp}_{2a}}) \oplus \operatorname{Std}_{\mathbf{SO}_{2b}^{\alpha}}) \circ \iota$, where we see α as a quadratic character of W_{F} .

2.4 Arthur's substitute for global parameters

In this section we let F be a number field. In order to formulate his multiplicity formula [Art13, Theorem 1.5.2], Arthur circumvented the absence of the hypothetical Langlands group by introducing substitutes for Arthur-Langlands parameters for special orthogonal and symplectic groups. Consider formal, unordered sums $\psi = \bigoplus_i \pi_i[d_i]$ where π_i is a selfdual automorphic cuspidal representation of \mathbf{GL}_{N_i}/F and $d_i \geq 1$ is an integer representing the dimension of an irreducible algebraic representation of $\mathbf{SL}_2(\mathbb{C})$. Let \mathbf{G} denote a quasisplit special orthogonal or symplectic group over F as in section 2.1, and let $N = N(\widehat{\mathbf{G}})$. Let $\widetilde{\Psi}_{\text{disc}}(\mathbf{G})$ be the set of $\psi = \bigoplus_i \pi_i[d_i]$ as above such that the pairs (π_i, d_i) are distinct, $\sum_i N_i d_i = N$, for all $i \operatorname{sign}(\pi_i)(-1)^{d_i-1} = \operatorname{sign}(\widehat{\mathbf{G}})$, and

- if **G** is symplectic, that is if $\widehat{\mathbf{G}}$ is odd orthogonal, $\prod_i \omega_{\pi_i}^{d_i} = 1$,
- if $\mathbf{G} = \mathbf{SO}_{2n}^{\alpha}, \prod_{i} \omega_{\pi_i}^{d_i} = \alpha$.

Note that if **G** is odd orthogonal, that is if $\widehat{\mathbf{G}}$ is symplectic, the condition $\prod_i \omega_{\pi_i}^{d_i} = 1$ is automatically satisfied.

Following Arthur, for a formal sum $\psi = \boxplus_i \pi_i [d_i]$ such that the pairs (π_i, d_i) are distinct we let \mathcal{L}_{ψ} be the fibre product of $({}^L\mathbf{G}_{\pi_i} \to \operatorname{Gal}_F)_i$. Recall that for each i we have $\operatorname{Std}_{\pi_i} : {}^L\mathbf{G}_{\pi_i} \to \operatorname{GL}_{N_i}(\mathbb{C})$, and so we can form the representation $\bigoplus_i \operatorname{Std}_{\pi_i} \otimes \nu_{d_i}$ of $\mathcal{L}_{\psi} \times \operatorname{SL}_2(\mathbb{C})$ where ν_d is the unique irreducible algebraic representation of $\operatorname{SL}_2(\mathbb{C})$ in dimension d. Then $\psi \in \widetilde{\Psi}_{\operatorname{disc}}(\mathbf{G})$ if and only there exists $\dot{\psi} : \mathcal{L}_{\psi} \times \operatorname{SL}_2(\mathbb{C}) \to {}^L\mathbf{G}$ such that $\operatorname{Std}_{\mathbf{G}} \circ \dot{\psi}$ is conjugated to $\bigoplus_i \operatorname{Std}_{\pi_i} \otimes \nu_{d_i}$. Moreover in this case $\dot{\psi}$ is unique up to conjugation by $\widehat{\mathbf{G}}$ except if \mathbf{G} is even orthogonal and all $N_i d_i$ are even, in which case there are exactly two conjugacy classes of such $\dot{\psi}$, exchanged by the group $\operatorname{Aut}({}^L\mathbf{G})/\widehat{\mathbf{G}}$ which has two elements. For $\psi \in \widetilde{\Psi}_{\operatorname{disc}}(\mathbf{G})$ let m_{ψ} be the number of $\widehat{\mathbf{G}}$ -conjugacy classes of such $\dot{\psi}$, so that $m_{\psi} \in \{1, 2\}$. Finally, denote by $\Psi_{\operatorname{disc}}(\mathbf{G})$ the set of such pairs $(\psi, \dot{\psi})$. To simplify the notation we will abusively denote by $\dot{\psi}$ such an element of $\Psi_{\operatorname{disc}}(\mathbf{G})$.

Let us recall some definitions from [Kot84, §10.2]. For $\dot{\psi} \in \Psi_{\text{disc}}(\mathbf{G})$ we let $C_{\dot{\psi}} = \text{Cent}(\dot{\psi}, \widehat{\mathbf{G}}) \supset Z(\widehat{\mathbf{G}})^{\text{Gal}_F}$. It is easy to check that $C_{\dot{\psi}}$ is a finite 2-group. We also let $S_{\dot{\psi}}$ be the group of $g \in \widehat{G}$ such that $g\dot{\psi}(x)g^{-1}\dot{\psi}(x)^{-1} \in Z(\widehat{\mathbf{G}})$ for all $x \in \mathcal{L}_{\psi} \times \text{SL}_2(\mathbb{C})$ and the resulting 1-cocycle $W_F \to Z(\widehat{\mathbf{G}})$ is locally trivial. In particular, $S_{\dot{\psi}}$ contains $Z(\widehat{\mathbf{G}})$. For the groups \mathbf{G} considered in this paper the action of Gal_F on $Z(\widehat{\mathbf{G}})$ factors through a cyclic extension, thus any locally trivial 1-cocycle $W_F \to Z(\widehat{\mathbf{G}})$ is trivial, and so $S_{\dot{\psi}} = C_{\dot{\psi}}Z(\widehat{\mathbf{G}})$. In fact the action of Gal_F on $Z(\widehat{\mathbf{G}})$ is trivial except in the uninteresting case $\mathbf{G} = \mathbf{SO}_2^{\alpha}$ with $\alpha \neq 1$, so that $S_{\dot{\psi}} = C_{\dot{\psi}}$ except in this case. Finally, let $S_{\dot{\psi}} = S_{\dot{\psi}}/Z(\widehat{\mathbf{G}})$. By the above remark, we also have $S_{\dot{\psi}} = C_{\dot{\psi}}/Z(\widehat{\mathbf{G}})^{\text{Gal}_F}$. The three groups $C_{\dot{\psi}}, S_{\dot{\psi}}$ and $S_{\dot{\psi}}$ can be canonically described using only ψ , in fact using only the family $(N_i d_i \mod 2)_i$ if \mathbf{G} is \mathbf{G}

fixed, and so to lighten our notations we will simply denote them by C_{ψ} , S_{ψ} and S_{ψ} in the sequel.

Let $(\mathbf{H}, \mathcal{H}, s, \xi)$ be an elliptic endoscopic datum for \mathbf{G} . Recall that in section 2.3 we fixed an L-isomorphism ${}^{L}\mathbf{H} \simeq \mathcal{H}$, and thus an embedding ${}^{L}\xi : {}^{L}\mathbf{H} \to {}^{L}\mathbf{G}$. Recall also that $\mathbf{H} = \mathbf{H}_{1} \times \mathbf{H}_{2}$ where \mathbf{H}_{1} and \mathbf{H}_{2} are also quasi-split special orthogonal or symplectic groups. Define $\widetilde{\Psi}_{\text{disc}}(\mathbf{H}) = \widetilde{\Psi}_{\text{disc}}(\mathbf{H}_{1}) \times \widetilde{\Psi}_{\text{disc}}(\mathbf{H}_{2})$ and similarly for $\Psi_{\text{disc}}(\mathbf{H})$. For $\psi' = (\psi'_{1}, \psi'_{2}) = (\boxplus_{i \in I_{1}} \pi_{i}[d_{i}], \boxplus_{i \in I_{2}} \pi_{i}[d_{i}]) \in \widetilde{\Psi}_{\text{disc}}(\mathbf{H})$ it is natural to define

$${}^{L}\xi(\psi') = \begin{cases} \boxplus_{i \in I_1 \sqcup I_2} \pi_i[d_i] & \text{if } \mathbf{G} = \mathbf{SO}_{2n+1} \text{ or } \mathbf{SO}_{2n}^{\alpha}, \\ (\boxplus_{i \in I_1}(\alpha \otimes \pi_i)[d_i]) \boxplus (\boxplus_{i \in I_2} \pi_i[d_i]) & \text{if } \mathbf{G} = \mathbf{Sp}_{2n}, \ \mathbf{H}_1 = \mathbf{Sp}_{2a} \text{ and } \mathbf{H}_2 = \mathbf{SO}_{2b}^{\alpha} \end{cases}$$

Clearly ${}^{L}\xi(\psi') \in \widetilde{\Psi}_{\text{disc}}(\mathbf{G})$ if and only if the pairs $(\pi_i, d_i)_{i \in I_1 \sqcup I_2}$ are distinct. Define $\widetilde{\Psi}_{\mathbf{G}-\text{disc}}(\mathbf{H}) \subset \widetilde{\Psi}_{\text{disc}}(\mathbf{H})$ as the subset of elements satisfying this condition, and define $\Psi_{\mathbf{G}-\text{disc}}(\mathbf{H})$ similarly. For $\psi' = (\psi'_1, \psi'_2) \in \widetilde{\Psi}_{\mathbf{G}-\text{disc}}(\mathbf{H})$ we can identify the fiber product of $\mathcal{L}_{\psi'_1} \to \text{Gal}_F$ and $\mathcal{L}_{\psi'_2} \to \text{Gal}_F$ with $\mathcal{L}_{L_{\xi}(\psi')}$. Note that in the case $\mathbf{G} = \mathbf{Sp}_{2n}$, this identification comes from identifying \mathbf{G}_{π_i} and $\mathbf{G}_{\alpha\otimes\pi_i}$ for $i \in I_1$. Thus we can define, for $\dot{\psi}' \in \Psi_{\mathbf{G}-\text{disc}}(\mathbf{H})$ above ψ' , its image ${}^{L}\xi(\dot{\psi}') \in \Psi_{\text{disc}}(\mathbf{G})$ above ${}^{L}\xi(\psi')$. Moreover $s \in S_{L_{\xi}(\dot{\psi}')}$.

Conversely, for $\dot{\psi} \in \Psi_{\text{disc}}(\mathbf{G})$, any $s \in S_{\psi}$ defines an elliptic endoscopic datum $(\mathbf{H}, \mathcal{H}, s, \xi)$ where $\xi(\widehat{\mathbf{H}}) = \text{Cent}(s, \widehat{\mathbf{G}})^0$ and $\xi(\mathcal{H}) = \xi(\widehat{\mathbf{H}})\dot{\psi}(\mathcal{L}_{\psi})$. The isomorphism class of $(\mathbf{H}, \mathcal{H}, s, \xi)$ clearly only depends on the image of s in \mathcal{S}_{ψ} , and there is a unique $\dot{\psi}' \in \Psi_{\mathbf{G}-\text{disc}}(\mathbf{H})$ such that $\dot{\psi} = {}^L \xi(\dot{\psi}')$. We obtain the following proposition.

Proposition 2.4.1 (Arthur). The mappings $\dot{\psi}' \mapsto {}^{L}\xi(\dot{\psi}')$ induce a bijection between isomorphism classes of pairs $((\mathbf{H}, \mathcal{H}, s, \xi), \dot{\psi}')$ where $(\mathbf{H}, \mathcal{H}, s, \xi)$ is an elliptic endoscopic datum for \mathbf{G} and $\dot{\psi}' \in \Psi_{\mathbf{G}-\text{disc}}(\mathbf{H})$, and $\widehat{\mathbf{G}}$ -conjugacy classes of pairs $(\dot{\psi}, s)$ where $\dot{\psi} \in \Psi_{\text{disc}}(\mathbf{G})$ and $s \in \mathcal{S}_{\psi}$.

Remark 2.4.2. The statement of the Proposition is particularly simple thanks to the fact that **G** and **H** satisfy the Hasse principle (see [Kot84, §4]). In general the "right" notion to consider is not that of a parameter $\dot{\psi}$ up to $\hat{\mathbf{G}}$ -conjugacy, but also up to locally trivial elements of $Z^1(W_F, Z(\hat{\mathbf{G}}))$, and similarly for **H**. Then an equivalence class of $\dot{\psi}'$ can correspond to more than one equivalence class of $(\dot{\psi}, s)$. We refer the interested reader to [Kot84, §11], particularly Proposition 11.2.1 for the general case.

2.5 Local Arthur packets in the quasi-split case

In this section we let F be a local field of characteristic zero, and \mathbf{G} one of the reductive groups over F defined in section 2.1. If F is p-adic, let $\mathcal{H}(\mathbf{G})$ be the Hecke algebra of smooth compactly supported distributions on \mathbf{G} with complex coefficients. If F is Archimedean, fix a maximal compact subgroup K of $\mathbf{G}(F)$ and let $\mathcal{H}(\mathbf{G})$ be the algebra of bi-K-finite smooth compactly supported distributions on $\mathbf{G}(F)$ with complex coefficients.

Contrary to the global case, we now have genuine Arthur-Langlands parameters at our disposal, i.e. continuous morphisms $\psi : WD_F \times SL_2(\mathbb{C}) \to {}^L\mathbf{G}$ such that

- the composition with ${}^{L}\mathbf{G} \to W_{F}$ is the obvious projection,
- for any $w \in WD_F$, $\psi(w)$ is semisimple,
- the restriction of ψ to the factor $SL_2(\mathbb{C})$ is algebraic.

We let $\Psi(\mathbf{G})$ be the set of Arthur-Langlands parameters for \mathbf{G} such that $\psi(WD_F)$ is bounded.

To formulate Arthur's construction of packets associated with such parameters precisely, it is necessary to fix a Whittaker datum \boldsymbol{w} for \mathbf{G} . If F is p-adic and \mathbf{G} is unramified, there is a unique $\mathbf{G}(F)$ -conjugacy class of hyperspecial maximal compact subgroups compatible with \boldsymbol{w} in the sense of [CS80]. In this case "unramified representation of $\mathbf{G}(F)$ " will mean unramified with respect to such a subgroup.

Arthur's construction relies on the local Langlands correspondence for general linear groups, via the standard representation $\operatorname{Std}_{\mathbf{G}}$ of ${}^{L}\mathbf{G}$. For this reason, the following fact is important: the Aut $({}^{L}\mathbf{G})$ -orbit of an Arthur-Langlands parameter ψ is determined by the conjugacy class of $\operatorname{Std}_{\mathbf{G}} \circ \psi$. Recall that for $\mathbf{G} = \mathbf{SO}_{2n}^{\alpha}$ the group $\operatorname{Out}({}^{L}\mathbf{G})$ has two elements, and so Arthur's methods using the twisted trace formula for general linear groups cannot distinguish between packets associated with $\widehat{\mathbf{G}}$ -conjugacy classes of parameters in the same $\operatorname{Out}({}^{L}\mathbf{G})$ -orbit. If $\mathbf{G} = \mathbf{SO}_{2n}^{\alpha}$, there exists an outer automorphism θ of \mathbf{G} which preserves \boldsymbol{w} : in fact θ can be realised as an element of the corresponding orthogonal group of determinant -1, as one can easily check for \mathbf{SO}_4^{α} , and the general case follows. If F is Archimedean, the maximal compact subgroup K of $\mathbf{G}(F)$ can be chosen to be stable under θ . Following Arthur, let $\mathcal{H}(\mathbf{G})$ be the subalgebra of $\mathcal{H}(\mathbf{G})$ consisting of θ -invariant distributions, so that irreducible representations of $\widetilde{\mathcal{H}}(\mathbf{G})$ correspond to $\{1, \theta\}$ orbits of irreducible representations. Note that it is not really necessary to consider θ invariant distributions, distributions whose orbital integrals are θ -invariant suffice: if π is an admissible representation of $\mathbf{G}(F)$ and $f(g)dg \in \mathcal{H}(\mathbf{G})$ has θ -invariant orbital integrals, then tr $(\pi(f(g)dg)) = \text{tr}(\pi^{\theta}(f(g)dg))$. If **G** is symplectic or odd-orthogonal, we simply let $\mathcal{H}(\mathbf{G}) = \mathcal{H}(\mathbf{G}) \text{ and } \theta = 1.$

As in the global case, denote by C_{ψ} the centraliser of ψ in $\widehat{\mathbf{G}}$, $S_{\psi} = Z(\widehat{\mathbf{G}})C_{\psi}$ and

$$\mathcal{S}_{\psi} = \pi_0 \left(C_{\psi} / Z(\widehat{\mathbf{G}})^{\operatorname{Gal}_F} \right) = \pi_0 \left(S_{\psi} / Z(\widehat{\mathbf{G}}) \right)$$

which is an abelian 2-group. Note that the image s_{ψ} of $-1 \in SL_2(\mathbb{C})$ by ψ belongs to C_{ψ} .

Arthur [Art13, Theorem 1.5.1] associates with any $\psi \in \Psi(\mathbf{G})$ a multi-set Π_{ψ} of $\{1, \theta\}$ orbits of irreducible unitary representations π of $\mathbf{G}(F)$, along with a map $\Pi_{\psi} \to \mathcal{S}_{\psi}^{\vee}$,
denoted by $\pi \mapsto \langle \cdot, \pi \rangle$. Arthur proves that the linear form

$$\Lambda_{\psi}: f(g)dg \mapsto \sum_{\pi \in \Pi_{\psi}} \langle s_{\psi}, \pi \rangle \operatorname{tr} \left(\pi(f(g)dg) \right)$$
(1)

is *stable*, i.e. vanishes if all stable orbital integrals of $f(g)dg \in \widetilde{\mathcal{H}}(\mathbf{G})$ vanish. These Arthur packets are characterised by [Art13, Theorem 2.2.1]. Note that our notation $\Lambda_{\psi}(f(g)dg)$ differs from Arthur's $f^{\mathbf{G}}(\psi)$.

Let us recall the endoscopic character relations [Art13, Theorem 2.2.1(b)]. Any semisimple $s \in S_{\psi}$ determines an endoscopic datum $(\mathbf{H}, \mathcal{H}, s, \xi)$ for \mathbf{G} . For any choice of Lembedding ${}^{L}\xi : {}^{L}\mathbf{H} \to {}^{L}\mathbf{G}$, there is a unique Arthur-Langlands parameter ψ' for \mathbf{H} such that $\psi = {}^{L}\xi \circ \psi'$. Note that \mathbf{H} is a product of general linear groups and quasi-split special orthogonal or symplectic groups, and so ψ' can be seen as a collection of Arthur-Langlands parameters for these groups. The endoscopic datum $(\mathbf{H}, \mathcal{H}, s, \xi)$, the L-embedding ${}^{L}\xi$ and the Whittaker datum \mathbf{w} allow to define *transfer factors* [KS99, §5.3], and thus to define the notion of distributions on $\mathbf{G}(F)$ and $\mathbf{H}(F)$ having matching orbital integrals ([KS99, §5.5], also called *transfer*). Thanks to the long-standing efforts of several mathematicians, any $f(g)dg \in \widetilde{\mathcal{H}}(\mathbf{G})$ is known to admit a transfer $f'(h)dh \in \widetilde{\mathcal{H}}(\mathbf{H})$. In our case, this follows from [Ng610], [Wal06] or [CHL11], [LS90, Theorem 2.3], [Wal95], and [Wal97]. Furthermore, [Hal95] implies that endoscopic transfer is compatible with the unramified Langlands correspondence. We refer the interested reader to [Cha09] for more information. We can now formulate the endoscopic character relation that holds for any s and f(g)dg:

$$\sum_{\pi \in \Pi_{\psi}} \langle s_{\psi}s, \pi \rangle \operatorname{tr}\left(\pi(f(g)dg)\right) = \Lambda_{\psi'}(f'(h)dh) = \sum_{\pi' \in \Pi_{\psi'}} \langle s_{\psi'}, \pi' \rangle \operatorname{tr}\left(\pi'(f'(h)dh)\right).$$
(2)

In a global setting it is also necessary to consider Arthur-Langlands parameters which are not bounded on WD_F , because the generalised Ramanujan conjecture is not known to hold in general. In this case the packet Π_{ψ} is simply defined by parabolic induction from the bounded case, so that its elements are not a priori irreducible. We refer to [Art13, §1.5] for details.

2.6 Localisation of global parameters

Finally, let us explain the relation between the previous two sections. Let F be a number field, and consider $(\psi, \dot{\psi}) \in \Psi_{\text{disc}}(\mathbf{G})$. Write ψ as $\boxplus_i \pi_i[d_i]$. Let v be a place of F. By [Art13, Theorem 1.4.2], for any i the Langlands parameter of the representation $\pi_{i,v}$ of $\mathbf{GL}_{N_i}(F_v)$ factors through $\operatorname{Std}_{\pi_i} : {}^L\mathbf{G}_{\pi_i} \to \operatorname{GL}_{N_i}(\mathbb{C})$. We obtain a Langlands parameter $\operatorname{WD}_{F_v} \to {}^L\mathbf{G}_{\pi}$, well-defined up to the action of $\operatorname{Aut}({}^L\mathbf{G}_{\pi_i})$. Composing with $\dot{\psi}$ we get an Arthur-Langlands parameter $\dot{\psi}_v : \operatorname{WD}_{F_v} \times \operatorname{SL}_2(\mathbb{C}) \to {}^L\mathbf{G}$. Moreover we have natural morphisms $C_{\dot{\psi}} \to C_{\dot{\psi}_v}, S_{\dot{\psi}} \to S_{\dot{\psi}_v}$ and $S_{\dot{\psi}} \to S_{\dot{\psi}_v}$.

3 Certain rigid inner forms and Arthur packets

3.1 Certain rigid inner forms

Let F be a totally real number field. From now on we use \mathbf{G}^* to denote a quasi-split special orthogonal or symplectic group over F as in section 2.1, as we will use \mathbf{G} to denote inner forms of \mathbf{G}^* . We shall be interested in certain inner twists of \mathbf{G}^* , i.e. pairs (\mathbf{G}, Ξ) where \mathbf{G} is a reductive group over F and $\Xi : \mathbf{G}_F^* \to \mathbf{G}_F$ is an isomorphism such that for any $\sigma \in \operatorname{Gal}_F$, the automorphism $\Xi^{-1}\sigma(\Xi)$ of \mathbf{G}_F^* is inner. For $\sigma \in \operatorname{Gal}_F$, denote by z_{σ} the element of $\mathbf{G}_{\mathrm{ad}}^*(\overline{F})$ such that $\Xi^{-1}\sigma(\Xi) = \operatorname{Ad}(z_{\sigma})$. This defines a cocyle z in $Z^1(F, \mathbf{G}_{\mathrm{ad}}^*)$, and the cohomology set $H^1(F, \mathbf{G}_{\mathrm{ad}}^*)$ classifies the isomorphism classes of inner twists of \mathbf{G}^* . Assume that F is not totally complex and let S be a non-empty set of real places of F. We say that (\mathbf{G}, S) satisfies property \star if

- for any place $v \in S$, the group $\mathbf{G}(F_v)$ admits discrete series,
- for any place $v \notin S$, the reductive group \mathbf{G}_{F_v} is quasi-split.

We will classify such pairs, and realise them as *rigid inner twists* (see [Kalb], [Kala]). In fact in the even orthogonal case we will restrict to *pure inner forms*, for reasons explained in Remark 3.1.1.

3.1.1 Rigid inner forms at real places

In this section we only consider a fixed real place v of F, and to simplify notation we denote without subscript the base change of \mathbf{G}^* to $F_v = \mathbb{R}$, and write $\operatorname{Gal}(\mathbb{C}/\mathbb{R}) = \{1, \tau\}$. We are interested in inner forms G of G^* . It is well-known that $G(\mathbb{R})$ admits discrete series if and only if $\mathbf{G}^*(\mathbb{R})$ admits discrete series, and this condition is also equivalent to \mathbf{G}^* admitting an anisotropic maximal torus, or to $\mathbf{G}^*(\mathbb{R})$ admitting an anisotropic inner form. Moreover there is at most one element of $H^1(\mathbb{R}, \mathbf{G}^*_{\mathrm{ad}})$ giving rise to an anisotropic inner form. Below we describe in which cases $\mathbf{G}^*(\mathbb{R})$ admits discrete series and recall the classification of its inner forms, restricting to *pure* inner forms in the even orthogonal case. In the orthogonal cases we will make these inner forms explicit by realising them as a *pure* inner forms. More precisely we will choose a pair $(\mathbf{B}^*, \mathbf{T}^*)$ where \mathbf{T}^* is an anisotropic maximal torus of \mathbf{G}^* and $\mathbf{B}^* \supset \mathbf{T}^*_{\mathbb{C}}$ is a Borel subgroup of $\mathbf{G}^*_{\mathbb{C}}$ such that all simple roots for $(\mathbf{B}^*, \mathbf{T}^*)$ are noncompact, and explicitly compute the surjective map $H^1(\mathbb{R}, \mathbf{T}^*) \to H^1(\mathbb{R}, \mathbf{G}^*)$. This will be useful in section 3.2 to compute the internal parametrisation of tempered discrete packets and Adams-Johnson packets for inner forms of \mathbf{G}^* . In the symplectic case the non-trivial inner forms of \mathbf{G}^* cannot be realised as pure inner forms, which is why the notion of rigid inner twist [Kalb] is needed. We denote by $u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}$ the extension of $\operatorname{Gal}(\mathbb{C}/\mathbb{R})$ that was denoted by $u \to W$ in [Kalb], to avoid confusion with the Weil group. We will fix a pair $(\mathbf{B}^*, \mathbf{T}^*)$ as in the orthogonal case and compute the surjective mappings

$$H^{1}(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z}(\mathbf{G}^{*}) \to \mathbf{T}^{*}) \to H^{1}(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z}(\mathbf{G}^{*}) \to \mathbf{G}^{*}) \to H^{1}(\mathbb{R}, \mathbf{G}_{\mathrm{ad}}^{*}).$$

Case G^{*} = SO_{2n+1} = SO(V,q). The group G^{*} is split and adjoint. The cohomology set H¹(ℝ, G^{*}) parametrises quadratic spaces having same dimension and discriminant as (V,q). Let T^{*} be an anisotropic maximal torus of G^{*}. It corresponds to an orthogonal decomposition (V,q) = (D,q₀) ⊕ ⊕ⁿ_{j=1}(P_j,q_j) where D is a line and each (P_j,q_j) is an anisotropic plane: T^{*} is the identity component of the stabilizer of this decomposition and T^{*} ≃ ∏_j SO(P_j,q_j). Up to replacing q by -q, we can assume that q₀ is positive, which is equivalent to disc(q) > 0. Then there are [n/2] negative (resp. [n/2] positive) planes among the P_j's. Thus up to reordering we can assume that (-1)^{n+1-j}q_j is positive for any j ≥ 1. For any j ≥ 1, fix a basis e_j of the free Z-module X^{*}(SO(P_j,q_j)) of rank one. There is a unique Borel subgroup B^{*} of G^{*}_C containing T^{*}_C such that the set of simple roots of T^{*}_C in B^{*} is

 $\{e_1 - e_2, \ldots, e_{n-1} - e_n, e_n\}$. This determines an identification $\widehat{\mathbf{T}^*} \simeq \mathcal{T}$ where \mathcal{T} is a maximal torus of $\widehat{\mathbf{G}^*} \simeq \operatorname{Sp}_{2n}(\mathbb{C})$, part of a pair $(\mathcal{B}, \mathcal{T})$ such that the set of simple roots of \mathcal{T} in \mathcal{B} is $\{e_1^* - e_2^*, \ldots, e_{n-1}^* - e_n^*, 2e_n^*\}$ where $e_j^*(e_k) = \delta_{j,k}$. The group $H^1(\mathbb{R}, \mathbf{T}^*)$ is in perfect duality with $\widehat{\mathbf{T}^*}^{\operatorname{Gal}(\mathbb{C}/\mathbb{R})}$ which is identified with the subgroup of 2-torsion in \mathcal{T} because τ acts by inversion on $\widehat{\mathbf{T}^*}$. The image of $(\epsilon_j)_{1\leq j\leq n} \in \{\pm 1\}^n \simeq H^1(\mathbb{R}, \mathbf{T}^*)$ in $H^1(\mathbb{R}, \mathbf{G}^*)$ corresponds to the isomorphism class of inner twists (\mathbf{G}, Ξ) where \mathbf{G} is a special orthogonal group of signature (1 + 2a, 2b), where

$$a = \operatorname{card} \{ j | \epsilon_j = (-1)^{n+1-j} \}, \ b = \operatorname{card} \{ j | \epsilon_j = (-1)^{n-j} \}.$$

In particular we see that the character of $Z(\widehat{\mathbf{G}^*})^{\operatorname{Gal}(\mathbb{C}/\mathbb{R})} = \{\pm 1\}$ that [Kot86, Theorem 1.2] associates to the image of $(\epsilon_j)_{1 \leq j \leq n}$ in $H^1(\mathbb{R}, \mathbf{G}^*)$ is $-1 \mapsto \prod_{1 \leq j \leq n} \epsilon_j = (-1)^{n(n-1)/2+a}$.

There is a unique element of $H^1(\mathbb{R}, \mathbf{T}^*)$ giving rise to the anisotropic inner form of \mathbf{G}^* , namely $((-1)^{n+1-j})_{1\leq j\leq n}$. The associated character of $Z(\widehat{\mathbf{G}^*})^{\operatorname{Gal}(\mathbb{C}/\mathbb{R})}$ is $-1 \mapsto (-1)^{n(n+1)/2}$.

• Case $\mathbf{G}^* = \mathbf{SO}_{2n}^{\alpha} = \mathbf{SO}(V,q)$, where $\alpha \in \mathbb{R}^{\times}/\mathbb{R}_{>0} \simeq \{\pm 1\}$ is equal to $(-1)^n \operatorname{disc}(q)$. The group \mathbf{G}^* admits discrete series if and only if $(-1)^n \alpha > 0$. Assume that this is the case. We choose q as in section 2.1, i.e. q has signature (n, n) (resp. (n+1, n-1)) if n is even (resp. odd). Let \mathbf{T}^* be an anisotropic maximal torus of \mathbf{G}^* . It corresponds to an orthogonal decomposition $(V,q) = \bigoplus_{j=1}^{n} (P_j,q_j)$ where each (P_j,q_j) is an anisotropic plane: \mathbf{T}^* is the identity component of the stabilizer of this decomposition and $\mathbf{T}^* \simeq \prod_j \mathbf{SO}(P_j, q_j)$. There are $\lceil n/2 \rceil$ positive (resp. $\lfloor n/2 \rfloor$ negative) planes among the (V_j, q_j) 's. Up to reordering we can assume that $(-1)^{j-1}q_j$ is positive definite for all j. As in the previous case fix bases e_i of $X^*(\mathbf{SO}(P_i, q_i))$, and let \mathbf{B}^* be the Borel subgroup corresponding to the set of simple roots $\{e_1 - e_2, \ldots, e_{n-1} - e_n\}$ $e_n, e_{n-1} + e_n$, and the set of simple roots for the pair $(\mathcal{B}, \mathcal{T})$ in $\widehat{\mathbf{G}^*} \simeq \mathrm{SO}_{2n}(\mathbb{C})$ is $\{e_1^* - e_2^*, \dots, e_{n-1}^* - e_n^*, e_{n-1}^* + e_n^*\}$. As before $H^1(\mathbb{R}, \mathbf{T}^*) \simeq \{\pm 1\}^n$ is in perfect duality with the subgroup of 2-torsion in \mathcal{T} . The set $H^1(\mathbb{R}, \mathbf{G}^*)$ parametrises the isomorphism classes of quadratic forms on dimension 2n vector spaces over \mathbb{R} having positive discriminant. The image of $(\epsilon_i)_{1 \le i \le n} \in \{\pm 1\}^n \simeq H^1(\mathbb{R}, \mathbf{T}^*)$ in $H^1(\mathbb{R}, \mathbf{G}^*)$ corresponds to the signature (2a, 2b) where

$$a = \operatorname{card} \{ j | \epsilon_j = (-1)^{j-1} \}, \ b = \operatorname{card} \{ j | \epsilon_j = (-1)^j \}$$

and the associated character of $Z(\widehat{\mathbf{G}^*})^{\operatorname{Gal}(\mathbb{C}/\mathbb{R})} = \{\pm 1\}$ is

$$-1 \mapsto \prod_{1 \le j \le n} \epsilon_j = (-1)^{n(n+1)/2+a}$$

Described using signatures, the fibres of the mapping $H^1(\mathbb{R}, \mathbf{G}^*) \to H^1(\mathbb{R}, \mathbf{G}^*_{\mathrm{ad}})$ are the sets $\{(2a, 2b), (2b, 2a)\}$ (having one or two elements).

There are two elements of $H^1(\mathbb{R}, \mathbf{T}^*)$ giving rise to anisotropic inner forms of \mathbf{G}^* , namely $((-1)^{j-1})_{1 \leq j \leq n}$ and $((-1)^j)_{1 \leq j \leq n}$. They map to distinct classes in $H^1(\mathbb{R}, \mathbf{G}^*)$, and the associated characters of $Z(\widehat{\mathbf{G}^*})^{\operatorname{Gal}(\mathbb{C}/\mathbb{R})}$ are $-1 \mapsto (-1)^{n(n-1)/2}$ and $-1 \mapsto (-1)^{n(n+1)/2}$. They map to the same class in $H^1(\mathbb{R}, \mathbf{G}^*_{\operatorname{ad}})$.

Case G^{*} = Sp_{2n} = Sp(V, a) where V is a 2n-dimensional vector space over ℝ and a is a non-degenerate alternating form. In this case G^{*} always admits discrete series. We have that H¹(ℝ, G^{*}) = 1 but for n > 0 the set H¹(ℝ, G^{*}_{ad}) is non-trivial. First we describe the anisotropic inner form as a rigid inner form. Choose J ∈ G^{*}(ℝ) such that J² = -1 and for any v ∈ V \{0\}, a(Jv, v) > 0. Choose i ∈ ℂ such that i² = -1. These define a complex structure on V, along with a positive definite hermitian form

$$h(v_1, v_2) := a(Jv_1, v_2) + ia(v_1, v_2).$$

Choose a decomposition $(V,h) = \bigoplus_{j=1}^{n} (P_j,h_j)$ where each P_j is a complex line. Then the stabiliser \mathbf{T}^* of this decomposition is an anisotropic maximal torus of \mathbf{G}^* . We have a canonical isomorphism $f_j : \mathbf{U}(P_j, h_j) \simeq \mathbf{U}_1$ for each j. Fix a basis b of $X^*(\mathbf{U}_1)$, and let $e_j = (-1)^{j-1} b \circ f_j$, a basis of $X^*(\mathbf{U}(P_j, h_j))$. Let $\mathbf{B}^* \supset \mathbf{T}^*_{\mathbb{C}}$ be the Borel subgroup of $\mathbf{G}^*_{\mathbb{C}}$ having as set of simple roots $\{e_1 - e_2, \ldots, e_{n-1} - e_n\}$ $e_n, 2e_n$. The basis (e_1, \ldots, e_n) identifies $X_*(\mathbf{T}^*)$ with \mathbb{Z}^n . Let $\mathbf{Z} \simeq \mu_2$ be the center of \mathbf{G}^* , and let $\overline{\mathbf{T}}^* = \mathbf{T}^*/\mathbf{Z}$, so that $X_*(\overline{\mathbf{T}}^*) = \mathbb{Z}^n + \mathbb{Z}(\frac{1}{2}, \ldots, \frac{1}{2})$. There are two elements of $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{T}^*) \simeq X_*(\overline{\mathbf{T}}^*)/2X_*(\mathbf{T}^*)$ ([Kalb, Theorem (4.8]) giving rise to the compact inner form of \mathbf{G}^* : $z_1 = \operatorname{cl}\left(\frac{1}{2}, -\frac{1}{2}, \ldots, \frac{(-1)^{n-1}}{2}\right)$ and $z_2 = \operatorname{cl}\left(-\frac{1}{2}, \frac{1}{2}, \ldots, \frac{(-1)^n}{2}\right)$. Note that $X_*(\overline{\mathbf{T}}^*)/2X_*(\mathbf{T}^*)$ is naturally isomorphic to the group of characters of the preimage of $\widehat{\mathbf{T}^*}^{\mathrm{Gal}(\mathbb{C}/\mathbb{R})}$ in $\widehat{\overline{\mathbf{T}^*}}$. The elements z_1, z_2 map to the same element in $H^1(\mathbb{R}, \overline{\mathbf{T}}^*) \simeq X_*(\overline{\mathbf{T}}^*)/2X_*(\overline{\mathbf{T}}^*)$, namely the one represented by $\tau \mapsto J$. Let (\mathbf{G}, Ξ) be a corresponding inner twist of \mathbf{G}^* . The positive definite quadratic form $v \mapsto a(Jv, v)$ on V yields a positive definite hermitian form h' on $\mathbb{C} \otimes_{\mathbb{R}} V$ seen as a complex vector space using the tensor product (not J!). It is easy to check that \mathbf{G} is a Zariski-closed subgroup of the anisotropic unitary group $\mathbf{U}(\mathbb{C}\otimes_{\mathbb{R}} V, h')$, and thus **G** is the anisotropic inner form of **G**^{*}. In fact we can give $\mathbb{C} \otimes_{\mathbb{R}} V$ a right \mathbb{H} -vector space structure by letting $j \in \mathbb{H}$ act by $\lambda \otimes v \mapsto \overline{\lambda} \otimes J(v)$, and **G** is the intersection of $\mathbf{U}(\mathbb{C}\otimes_{\mathbb{R}} V, h')$ with the real algebraic group of automorphisms of this H-vector space.

Recall [Kalb, §3.4] the quotient $H^1_{\rm sc}(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{G}^*)$ of the pointed set $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{G}^*)$. This quotient has a natural structure of abelian group. By [Kalb, Theorem 4.11 and Proposition 5.3] (which generalise [Kot86, Theorem 1.2]), $H^1_{\rm sc}(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{G}^*) \simeq Z(\widehat{\mathbf{G}^*}_{\rm sc})^{\vee} \simeq \mathbb{Z}/2\mathbb{Z}$, and the image of z_i is the non-trivial element, for any *i*. For any *i*, the fibre of this non-trivial element can be identified with $\Xi^{-1}(H^1(\mathbb{R}, \mathbf{G})) z_i$, and thus

$$H^{1}(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{G}^{*}) = \{1\} \bigsqcup \Xi^{-1} \left(H^{1}(\mathbb{R}, \mathbf{G}) \right) z_{i}.$$

The pointed set $H^1(\mathbb{R}, \mathbf{G})$ can be computed using [Ser94, Chapitre III, Théorème 6], and it is in natural bijection with the set of conjugacy classes of elements y of

order 1 or 2 in $\mathbf{G}(\mathbb{R})$ (equivalently, $\mathbf{G}(\mathbb{C})$). Mapping y to dim ker $(y + \mathrm{Id})/2$ (in the standard representation of $\mathbf{G}(\mathbb{C})$) allows to identify $H^1(\mathbb{R}, \mathbf{G})$ with the integer interval $\{0, \ldots, n\}$. The trivial element of $H^1(\mathbb{R}, \mathbf{G})$ corresponds to 0 and the element x such that $\Xi^{-1}(x)z_i = z_{3-i}$ corresponds to n. The elements of $\Xi^{-1}(H^1(\mathbb{R}, \mathbf{G})) z_i$ corresponding to integers k and k' map to the same class in $H^1(\mathbb{R}, \mathbf{G}_{\mathrm{ad}})$ if and only if $k' \in \{k, n - k\}$. The trivial class in $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{G}^*)$ is the only one mapping to the trivial class in $H^1(\mathbb{R}, \mathbf{G}_{\mathrm{ad}}^*)$. In particular $H^1(\mathbb{R}, \mathbf{G}_{\mathrm{ad}}^*)$ has $\lfloor n/2 \rfloor + 2$ elements.

Remark 3.1.1. The reader might wonder why we have not considered rigid inner forms instead of only pure inner forms in the even orthogonal case. For $n \ge 2$ denote $\mathbf{G}^* =$ $\mathbf{SO}_{2n}^{(-1)^n}$. It is not difficult to compute that the complement of $H^1(\mathbb{R}, \mathbf{G}^*)$ in $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z}(\mathbf{G}^*) \to \mathbf{G}^*)$ has two elements. They map to the same element in $H^1(\mathbb{R}, \mathbf{G}_{\mathrm{ad}}^*)$ if and only if n is odd. These two elements are swapped by the outer automorphism of \mathbf{G}^* . For this reason we cannot formulate Proposition 3.2.6 for inner forms of even special orthogonal groups which are not pure inner forms, and thus we cannot formulate Theorem 4.0.1 either. Note that for n even, any non-pure inner form of \mathbf{SO}_{2n}^1 does not admit any outer automorphism defined over \mathbb{R} .

3.1.2 Global rigid inner forms

We can now classify which groups among the quasi-split groups \mathbf{G}^* over a number field F defined in section 2.1 admit inner forms \mathbf{G} satisfying property \star with respect to a given non-empty set S of real places of F. As explained in the previous remark, we restrict ourselves to *pure* inner forms in the even orthogonal case. Since odd special orthogonal groups are adjoint, we only have to consider pure inner forms when \mathbf{G}^* is orthogonal. In the symplectic case we will realise all inner forms \mathbf{G} of \mathbf{G}^* satisfying property \star with respect to S as *rigid inner forms*. We refer to [Kalb] (resp. [Kala]) for the construction of the extension $u_v \to \mathcal{E}_v$ of Gal_{F_v} (resp. $P_{F,\dot{V}} \to \mathcal{E}_{F,\dot{V}}$ of Gal_F) by a profinite algebraic group, and the cohomology sets for reductive groups defined using these extensions. Note that as in the previous section we denote by $u_v \to \mathcal{E}_v$ the extension that is denoted by $u_v \to W_v$ in [Kalb], to avoid confusion with the Weil group, and that we suppressed the choice of $\tilde{\xi}$ in the notation $\mathcal{E}_{F\dot{V}}$. Let \mathbf{Z} denote the center of \mathbf{G}^* .

Of course up to removing places from S we only have to consider groups \mathbf{G}^* that are not quasi-split at all places in S.

Proposition 3.1.2. Let F be a totally real number field, S a non-empty set of real places of F, and $n \ge 1$.

1. $\mathbf{G}^* = \mathbf{SO}_{2n+1} = \mathbf{SO}(V, q)$ where q has positive discriminant at all places of S. For each place $v \in S$ choose $z_v \in H^1(F_v, \mathbf{G}^*)$, corresponding to a signature $(1 + 2a_v, 2b_v)$ where $a_v + b_v = n$, with associated character of $Z(\widehat{\mathbf{G}^*})^{\operatorname{Gal}_{F_v}} \simeq \{\pm 1\}$:

$$-1 \mapsto \epsilon_v = (-1)^{n(n-1)/2 + a_v}$$

Then \mathbf{G}^* admits an inner form satisfying property \star with respect to S which is isomorphic to z_v at all places $v \in S$ if and only if $\prod_{v \in S} \epsilon_v = 1$. If such an inner form

exists then it is unique up to isomorphism.

2. $\mathbf{G}^* = \mathbf{SO}_{2n}^{\alpha}$ where $\alpha \in F^{\times}/F^{\times 2}$ and $\mathbf{G}^* = \mathbf{SO}(V,q)$ where q is defined in section 2.1. Assume that $(-1)^n \alpha$ is positive at all places in S. For each $v \in S$ choose $z_v \in H^1(F_v, \mathbf{G}^*)$, corresponding to the signature $(2a_v, 2b_v)$ where $a_v + b_v = n$ and with associated character of $Z(\widehat{\mathbf{G}^*})^{\operatorname{Gal}_{F_v}} \simeq \{\pm 1\}$:

$$-1 \mapsto \epsilon_v = (-1)^{n(n+1)/2 + a_v}$$

(a) Assume that n is odd. Then G* has a unique inner form G satisfying property
 ★ with respect to S and isomorphic to the inner form given by the image of z_v in H¹(F_v, G^{*}_{ad}) at all v ∈ S.

Let V be a finite set of real or finite places of F disjoint from S and such that for any $v \in V$, $\alpha \notin F_v^{\times 2}$, and moreover $(-1)^{\operatorname{card}(V)} = \prod_{v \in S} \epsilon_v$. For any $v \in V$ let $z_v \in H^1(F_v, \mathbf{G}^*)$ be the unique non-trivial class mapping to the trivial class in $H^1(F_v, \mathbf{G}^*_{\operatorname{ad}})$. Explicitly, z_v corresponds to the signature (n-1, n+1) if v is real, and z_v is the unique non-trivial element of $H^1(F_v, \mathbf{G}^*)$ if v is finite. Then there is a unique $z \in H^1(F, \mathbf{G}^*)$ such that for any $v \in S \cup V$ the image of z in $H^1(F_v, \mathbf{G}^*)$ equals z_v , and for any $v \notin S \cup V$ the image of z in $H^1(F_v, \mathbf{G}^*)$ is trivial. The corresponding pure inner form of \mathbf{G}^* is isomorphic to \mathbf{G} as an inner form, and this describes all the pure inner forms of \mathbf{G}^* satisfying this property.

(b) Assume that n is even. Then \mathbf{G}^* has an inner form \mathbf{G} satisfying property \star with respect to S and isomorphic to the image of z_v in $H^1(F_v, \mathbf{G}^*_{\mathrm{ad}})$ at all places $v \in S$ if and only if $\prod_{v \in S} \epsilon_v = 1$ or $\alpha \neq 1$. If it exists then it is unique up to isomorphism.

Assume that $\prod_{v \in S} \epsilon_v = 1$ or $\alpha \neq 1$. Let V be a finite set of real or finite places of F disjoint from S and such that for any $v \in V$, $\alpha \notin F_v^{\times 2}$, and moreover $(-1)^{\operatorname{card}(V)} = \prod_{v \in S} \epsilon_v$. Such a set exists and $V = \emptyset$ is the only possible choice if $\alpha = 1$. For any $v \in V$ let $z_v \in H^1(F_v, \mathbf{G}^*)$ be the unique non-trivial class mapping to the trivial class in $H^1(F_v, \mathbf{G}^*)$. Explicitly, z_v corresponds to the signature (n - 1, n + 1) if v is real, and z_v is the unique non-trivial element of $H^1(F_v, \mathbf{G}^*)$ if v is finite. Then there is a unique $z \in H^1(F, \mathbf{G}^*)$ such that for any $v \in S \cup V$ the image of z in $H^1(F_v, \mathbf{G}^*)$ equals z_v , and for any $v \notin S \cup V$ the image of z in $H^1(F_v, \mathbf{G}^*)$ is trivial. The corresponding pure inner form of \mathbf{G}^* is isomorphic to \mathbf{G} as an inner form, and this describes all the pure inner forms of \mathbf{G}^* satisfying this property.

3. $\mathbf{G}^* = \mathbf{Sp}_{2n}$. For each $v \in S$ choose $\overline{z}_v \in H^1(F_v, \mathbf{G}^*_{ad}) \smallsetminus \{1\}$. Then \mathbf{G}^* has an inner form satisfying property \star with respect to S and which is isomorphic to the image of \overline{z}_v at each place $v \in S$ if and only if $\operatorname{card}(S)$ is even. If it exists then it is unique up to isomorphism.

Let **Z** be the center of **G**^{*}. For each $v \in S$ choose $z_v \in H^1(u_v \to \mathcal{E}_v, \mathbf{Z} \to \mathbf{G}^*) \setminus \{1\}$ lifting \overline{z}_v . If card(S) is even, then up to isomorphism there is a unique global rigid inner form which is isomorphic to z_v at any $v \in S$ and split at all places of F not in S.

Proof. Note that in all cases the groups $\mathbf{G}_{\mathrm{ad}}^*$ and \mathbf{G}^* and all their inner forms satisfy the Hasse principle by [Kot84, §4], since the action of Gal_F on the center of the dual group factors through a cyclic group. This fact implies all claims of unicity appearing in the Proposition.

The first two cases of special orthogonal groups are simple exercises using the previous section, [Kot86] and [PR94, Theorem 6.6], as is the first part of the symplectic case. For the last part of the third case we need Kaletha's generalisation of Kottwitz' results. Assume that $\mathbf{G}^* = \mathbf{Sp}_{2n}$ and that $\operatorname{card}(S)$ is even. Note that for $v \notin S$, the trivial class in $H^1(u_v \to \mathcal{E}_v, \mathbf{Z} \to \mathbf{G}^*)$ is the only one giving rise to a split inner form of $\mathbf{G}_{F_u}^*$: this is obvious if v is complex, was observed in the previous section for real v, and finally for a finite place v we simply have $H^1(u_v \to \mathcal{E}_v, \mathbf{Z} \to \mathbf{G}^*) \simeq H^1(F_v, \mathbf{G}^*_{\mathrm{ad}})$ because $\mathbf{G}^*_{F_v}$ is split and semisimple (see [Kalb, Proposition 5.3]). By [Kala, Corollary 3.45], describing the image of the localisation mapping $H^1(P_{F,\dot{V}} \to \mathcal{E}_{F,\dot{V}}, \mathbf{Z} \to \mathbf{G}^*) \to \prod_v H^1(u_v \to \mathcal{E}_v, \mathbf{Z} \to \mathbf{G}^*),$ there is at least one relevant $z \in H^1(P_{F,\dot{V}} \to \mathcal{E}_{F,\dot{V}}, \mathbf{Z} \to \mathbf{G}^*)$. Using [Kala, Lemma 3.22] we see that the image of z by $H^1(P_{F,\dot{V}} \to \mathcal{E}_{F,\dot{V}}, \mathbf{Z} \to \mathbf{G}^*) \to \operatorname{Hom}(P_{\dot{V}}, \mathbf{Z})$ does not depend on the choice of a relevant z, and so two such relevant classes z_1, z_2 differ by an element of $H^1(F, \mathbf{G}_1) \simeq \prod_{v \in S} H^1(F_v, \mathbf{G}_1)$, where \mathbf{G}_1 is the inner form of \mathbf{G}^* determined by z_1 . Since twisting by $z_{1,v}$ embeds $H^1(F_v, \mathbf{G}_1)$ into $H^1(u_v \to \mathcal{E}_v, \mathbf{Z} \to \mathbf{G}^*)$ for all $v \in S$, we have that $z_1 = z_2$.

Note that pure inner forms are special cases of rigid inner forms by letting $\mathbf{Z} = 1$, thus all cases could be formulated using rigid inner forms. It is not difficult to check that any inner form of an even special orthogonal group which is anisotropic at all Archimedean places of F and quasi-split at all finite places of F arises as a pure inner forms, i.e. is a special orthogonal group. In the symplectic case the inner forms considered in the Proposition can be constructed explicitly as unitary groups over a quaternion algebra over F which is non-split exactly at the places in S.

Remark 3.1.3. We will use this realisation of an inner form \mathbf{G} as a rigid inner form of \mathbf{G}^* to get a coherent family of normalisations of the local Langlands correspondences for the groups $\mathbf{G}(F_v)$, following [Kalb]. Here "coherent" can be intuitively understood as "satisfying a product formula", a condition that is necessary to formulate the main goal of this paper, i.e. Arthur's multiplicity formula for automorphic representations of \mathbf{G} (Theorem 4.0.1). In the case $\mathbf{G}^* = \mathbf{SO}_{2n+1}$ (resp. $\mathbf{G}^* = \mathbf{Sp}_{2n}$) for any $z \in H^1(F, \mathbf{G}^*)$ (resp. $H^1(P_{F,\dot{V}} \to \mathcal{E}_{F,\dot{V}}, \mathbf{Z} \to \mathbf{G}^*)$) as in the Proposition and any place $v \notin S$, the image of z in $H^1(F_v, \mathbf{G}^*)$ (resp. $H^1(u_v \to \mathcal{E}_v, \mathbf{Z} \to \mathbf{G}^*)$) is trivial. This is not always the case for $\mathbf{G}^* = \mathbf{SO}_{2n}^2$. In fact it can happen that no relevant $z \in H^1(F, \mathbf{G}^*)$ is locally trivial at all places of F not in S, i.e. that V as in the Proposition is necessarily non-empty: this occurs when $\prod_{v \in S} \epsilon_v = -1$. In these cases it is not possible to rely solely on Arthur's construction of packets reviewed in section 2.5. Fortunately Kaletha calculated the effect of twisting the

local Langlands correspondence by an element of $H^1(F_v, \mathbf{Z}(\mathbf{G}))$, and we will review his results in section 3.3.

3.2 Local Arthur packets for real reductive groups having discrete series

Let \mathbf{G}^* be a quasi-split reductive group as in section 2.1 over \mathbb{R} , and assume that \mathbf{G}^* admits discrete series. Let \mathbf{G} be an inner form of \mathbf{G}^* . Let \mathbf{T}^* be an anisotropic maximal torus of \mathbf{G}^* , and let \mathbf{Z} be the center of \mathbf{G}^* , except for $\mathbf{G}^* = \mathbf{SO}_2^{\alpha}$ (with $\alpha = -1$) which is a torus, in which case we let \mathbf{Z} be any finite subgroup of \mathbf{G}^* . As we saw in section 3.1.1, the inner form \mathbf{G} of \mathbf{G}^* can be realised as a rigid inner twist (\mathbf{G}, Ξ, z) of \mathbf{G}^* , i.e. $\Xi :$ $\mathbf{G}^*_{\mathbb{C}} \to \mathbf{G}_{\mathbb{C}}$ is an isomorphism and $z \in Z^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{G}^*)$ is such that for any $w \in \mathcal{E}_{\mathbb{R}}$, $\Xi^{-1}w(\Xi) = \mathrm{Ad}(z(w))$. In fact we realised z as an element of $Z^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{T}^*)$. Note that for now we do not assume that \mathbf{G} is a pure inner form in the even orthogonal case, but we will have to make this assumption again in section 3.2.3.

According to [Kalb, §5.6], which makes precise certain constants in Shelstad's work ([She08a], [She10], [She08b]), this realisation and the choice of a Whittaker datum for \mathbf{G}^* are enough to normalise the local Langlands correspondence for \mathbf{G} , that is to determine an internal parametrisation of each L-packet $\Pi_{\varphi}(\mathbf{G})$ of representations of $\mathbf{G}(\mathbb{R})$ corresponding to a tempered Langlands parameter $W_{\mathbb{R}} \to {}^L \mathbf{G}$, in a manner analogous to the quasi-split case reviewed in section 2.5. In the anisotropic case, the L-packet $\Pi_{\varphi}(\mathbf{G})$ is a singleton.

In the global setting we will also need to consider non-tempered Arthur-Langlands parameters, i.e. Arthur-Langlands parameters $W_{\mathbb{R}} \times \operatorname{SL}_2(\mathbb{C}) \to {}^L \mathbf{G}$ whose restriction to $\operatorname{SL}_2(\mathbb{C})$ is non-trivial. For the main result of this paper (Theorem 4.0.1) we will restrict to Arthur-Langlands parameters having algebraic regular infinitesimal character, i.e. those considered by Adams and Johnson in [AJ87]. Adams and Johnson worked with arbitrary real reductive groups (not necessarily quasi-split) and proved endoscopic character relations similar to (2), but only up to a multiplicative constant. We will have to check that Kaletha's definitions allow to remove this ambiguity (Proposition 3.2.5), similarly to what we did in [Taïa, §4.2.2] for the quasi-split case. We will see that in the anisotropic case, these Adams-Johnson packets are singletons.

Recently Arancibia, Moeglin and Renard [AMR] proved that the packets of representations of $\mathbf{G}^*(\mathbb{R})$ constructed by Adams and Johnson coincide with those constructed by Arthur in [Art13], compatibly with the internal parametrisations. We will show (Proposition 3.2.6) that this implies endoscopic character relations between Adams-Johnson packets for \mathbf{G} and Arthur packets reviewed in section 2.5 for endoscopic groups of \mathbf{G} . The only difficulty comes from the fact that Arthur packets are sets of representations up to outer automorphism. For this reason in the case $\mathbf{G}^* = \mathbf{SO}_{2n}^{\alpha}$ we will restrict ourselves to pure inner forms \mathbf{G} . Of course this covers the case of anisotropic inner forms in which we are particularly interested.

3.2.1 Tempered parameters

We begin by recalling Shelstad and Kaletha's internal parametrisation of L-packets for rigid inner twists of quasi-split groups [Kalb, §5.6]. We adopt a formulation slightly differing

from [Kalb] and closer to [Taïa, §4.2.1], to prepare for the non-tempered case. We will be brief and focus on the aspects which were not examined in [Taïa, $\S4.2.1$]. It is convenient to let the inner form \mathbf{G} of \mathbf{G}^* vary. Note that since \mathbf{G} and \mathbf{G}^* are inner forms of each other, their Langlands dual groups are canonically identified, and we will only use the notation ^LG. Recall that by definition ([Bor79]) $\widehat{\mathbf{G}}$ comes with a pinning $(\mathcal{B}, \mathcal{T}, (\mathcal{X}_{\alpha})_{\alpha \in \Delta})$, used to define the action of $W_{\mathbb{R}}$ on $\widehat{\mathbf{G}}$ and ${}^{L}\mathbf{G} = \widehat{\mathbf{G}} \rtimes W_{\mathbb{R}}$. The $\widehat{\mathbf{G}}$ -conjugacy classes of *discrete* Langlands parameters $\varphi_{\lambda}: W_{\mathbb{R}} \to {}^{L}\mathbf{G}^{*}$ are parametrised by dominant weights λ for \mathbf{G} . We can assume that φ_{λ} is aligned with $(\mathcal{B}, \mathcal{T})$ in the sense that $\varphi_{\lambda}(z) = (2\tau)(z/|z|) \rtimes z$ for $z \in W_{\mathbb{C}} = \mathbb{C}^{\times}$, where $\tau = \lambda + \rho$ (with 2ρ equal to the sum of the positive roots) is seen as an element of $\frac{1}{2}\mathbb{Z}\otimes X_*(\mathcal{T})$, strictly dominant for \mathcal{B} . See [Lan89, Lemma 3.2] for the extension of φ_{λ} to $W_{\mathbb{R}}$. Since τ is strictly dominant, $C_{\varphi_{\lambda}} = \{t \in \mathcal{T} \mid t^2 = 1\}$. Denote by $C_{\varphi_{\lambda}}^+$ the preimage of $C_{\varphi_{\lambda}}$ in $\widehat{\overline{\mathbf{G}}}$ where $\overline{\mathbf{G}} = \mathbf{G}/\mathbf{Z}$. Denote by $Z(\widehat{\overline{\mathbf{G}}})^+$ the preimage of $Z(\widehat{\mathbf{G}})^{\operatorname{Gal}(\mathbb{C}/\mathbb{R})}$ in $Z(\widehat{\overline{\mathbf{G}}})$. Then $Z(\widehat{\overline{\mathbf{G}}})^+ \subset C_{\varphi_{\lambda}}^+$. Let $\mathcal{S}_{\varphi_{\lambda}}^+ = \pi_0(C_{\varphi_{\lambda}}^+)$, so that there is a surjective morphism $\mathcal{S}_{\varphi_{\lambda}}^{+} \to \mathcal{S}_{\varphi_{\lambda}}$ whose kernel is the image of $Z(\widehat{\overline{\mathbf{G}}})^{+}$ in $\mathcal{S}_{\varphi_{\lambda}}^{+}$. We have that $C_{\varphi_{\lambda}}^{+}$ has dimension 0 and so $\mathcal{S}_{\varphi_{\lambda}}^{+} = C_{\varphi_{\lambda}}^{+}$ and the above kernel is $Z(\widehat{\overline{\mathbf{G}}})^{+}$. Note also that $C_{\varphi_{\lambda}}^{+}$ is contained in a maximal torus of $\overline{\mathbf{G}}$, and thus $\mathcal{S}_{\varphi_{\lambda}}^{+}$ is abelian.

For any inner form **G** of **G**^{*}, the L-packet $\Pi_{\varphi_{\lambda}}(\mathbf{G})$ of discrete series representations of $\mathbf{G}(\mathbb{R})$ associated with φ_{λ} is in bijection with the set $\Sigma_{\mathcal{B}}(\mathbf{G})$ of $\mathbf{G}(\mathbb{R})$ -conjugacy classes of pairs (\mathbf{B}, \mathbf{T}) where \mathbf{T} is an anisotropic maximal torus of \mathbf{G} and \mathbf{B} is a Borel subgroup of $\mathbf{G}_{\mathbb{C}}$ containing $\mathbf{T}_{\mathbb{C}}$. This is a consequence of Harish-Chandra's formulae for the characters of discrete series. Choose a Whittaker datum \boldsymbol{w} for \mathbf{G}^* . By work of Kostant and Vogan, for any dominant weight λ there is a unique representation in $\Pi_{\varphi_{\lambda}}(\mathbf{G}^{*})$ which is generic for \boldsymbol{w} . It corresponds to a $\mathbf{G}^*(\mathbb{R})$ -conjugacy class $\operatorname{cl}(\mathbf{B}^*, \mathbf{T}^*)$ which does not depend on λ and has the property that all the simple roots of $\mathbf{T}^*_{\mathbb{C}}$ in \mathbf{B}^* are non-compact. There is a bijection between $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{T}^*)$ and the set of isomorphism classes of $(\mathbf{G}, \Xi, z, \mathbf{B}, \mathbf{T})$ where (\mathbf{G}, Ξ, z) is a rigid inner twist of \mathbf{G}^* and (\mathbf{B}, \mathbf{T}) is a pair in \mathbf{G} as above, obtained by mapping $z \in Z^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbb{Z} \to \mathbb{T}^*)$ to $(\mathbf{G}, \Xi, z, \Xi(\mathbf{B}^*), \Xi(\mathbf{T}^*))$, for an arbitrary choice of inner form (\mathbf{G}, Ξ) compatible with z. The surjectivity essentially comes from the fact that any automorphism of $\mathbf{T}^*_{\mathbb{C}}$ is defined over \mathbb{R} . This gives a bijection between $H^1(u_{\mathbb{R}} \to$ $\mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{T}^*$) and the set of isomorphism classes of pairs $((\mathbf{G}, \Xi, z), \pi)$ where (\mathbf{G}, Ξ, z) is a rigid inner twist of \mathbf{G}^* and $\pi \in \Pi_{\varphi_{\lambda}}(\mathbf{G})$. Finally, the choice of \mathbf{B}^* allows to identify \mathcal{T} with $\widehat{\mathbf{T}^*}$, and thus to identify $(\mathcal{S}_{\varphi_{\lambda}}^+)^{\vee}$ with $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{T}^*)$, using [Kalb][Corollary 5.4]. This concludes the construction of the mapping $\Pi_{\varphi_{\lambda}}(\mathbf{G}) \to (\mathcal{S}^+_{\varphi_{\lambda}})^{\vee}$, for any rigid inner twist (\mathbf{G}, Ξ, z) of \mathbf{G}^* . We will abusively denote this mapping by $\pi \mapsto \langle \cdot, \pi \rangle$, although it depends on the realisation of \mathbf{G} as a rigid inner twist of \mathbf{G}^* and the choice of the Whittaker datum \mathfrak{w} . Shelstad and Kaletha also prove endoscopic character relations, analogous to equation (2). Fix the dominant weight λ and denote $\varphi = \varphi_{\lambda}$ for simplicity. Any element $\dot{s} \in \mathcal{S}_{\varphi}^{+}$ yields a *refined endoscopic datum*, a notion defined in [Kalb, §5.3], $\dot{\mathfrak{e}} = (\mathbf{H}, \mathcal{H}, \dot{s}, \xi)$ for \mathbf{G}^* , along with a parameter $\varphi': W_{\mathbb{R}} \to \mathcal{H}$ such that $\varphi = \xi \circ \varphi'$. Moreover the underlying endoscopic datum is elliptic. Choose an L-embedding $^{L}\xi$ for this endoscopic datum, which is equivalent to choosing an isomorphism ${}^{L}\mathbf{H} \simeq \mathcal{H}$. This allows us to see φ' as a parameter for **H**, and φ' is clearly discrete. For any rigid inner twist (\mathbf{G}, Ξ, z) of \mathbf{G}^* , Kaletha defines absolute transfer factors $\Delta'[\dot{\mathbf{e}}, {}^L\xi, \mathbf{w}, \Xi, z]$. We replaced the z-pair denoted by \mathfrak{z} in [Kalb] by ${}^L\xi$, as we argued in section 2.3 that it is not necessary to introduce a z-extension of **H**. By [Kalb, Proposition 5.10], if f(g)dg and f(h)dh are $\Delta'[\dot{\mathbf{e}}, {}^L\xi, \mathbf{w}, \Xi, z]$ -matching smooth compactly supported distributions on $\mathbf{G}(\mathbb{R})$ and $\mathbf{H}(\mathbb{R})$, we have

$$e(\mathbf{G})\sum_{\pi\in\Pi_{\varphi}(\mathbf{G})}\langle \dot{s},\pi\rangle\mathrm{tr}\left(\pi(f(g)dg)\right) = \sum_{\pi'\in\Pi_{\varphi'}(\mathbf{H})}\mathrm{tr}\left(\pi'(f'(h)dh)\right).$$
(3)

Here $e(\mathbf{G}) = (-1)^{q(\mathbf{G})-q(\mathbf{G}^*)}$ is the sign defined by Kottwitz in [Kot83].

Remark 3.2.1. The restriction of each $\langle \cdot, \pi \rangle$ to $Z(\widehat{\mathbf{G}})^+$ coincides with the image of z by [Kalb][Corollary 5.4], which justifies the use of the transfer factors Δ' rather than Δ (see [KS]). Note that in section 2.5 Δ was used, but we could have used Δ' as well since for any endoscopic datum considered in this paper we have $s^2 = 1$.

Remark 3.2.2. For an anisotropic rigid inner twist (\mathbf{G}, Ξ, z) of \mathbf{G}^* , there is a unique $\mathbf{G}(\mathbb{R})$ -conjugacy class of pairs (\mathbf{B}, \mathbf{T}) in \mathbf{G} as above. Therefore for any discrete parameter φ the L-packet $\Pi_{\varphi}(\mathbf{G})$ consists of only one representation.

Example 3.2.3. Let us end this section by discussing an example. For $n \ge 2$ let $\mathbf{G}^* = \mathbf{SO}_{2n}^{(-1)^n} = \mathbf{SO}(V,q)$ where q has signature (n,n) (resp. (n+1,n-1)) if n is even (resp. odd). We use the same notation as in section 3.1.1. If n is odd the quasi-split group \mathbf{G}^* has a unique $\mathbf{G}^*(\mathbb{R})$ -conjugacy class of Whittaker datum \mathfrak{w} . If n is even there are two such conjugacy classes, and we let \mathfrak{w} be the one corresponding to the pair $(\mathbf{B}^*, \mathbf{T}^*)$ chosen in section 3.1.1. Let (\mathbf{G}, Ξ, z) where $z \in Z^1(\mathbb{R}, \mathbf{T}^*)$ be the pure inner twist of \mathbf{G}^* corresponding to the signature (2n, 0).

We realise the dual group ${}^{L}\mathbf{G} = {}^{L}\mathbf{G}^{*}$ as in section 2.1, using the symmetric bilinear form $B(v_{i}, v_{j}) = \delta_{i,2n+1-j}$ on $\mathbb{C}v_{1} \oplus \cdots \oplus v_{2n}$. Recall that the set of simple roots for $(\mathbf{B}^{*}, \mathbf{T}^{*})$ is $\Delta = \{e_{1} - e_{2}, \ldots, e_{n-1} - e_{n}, e_{n-1} + e_{n}\}$. Let \mathcal{T} be the maximal torus of $\widehat{\mathbf{G}}$ consisting of diagonal matrices $t = \operatorname{diag}(t_{1}, \ldots, t_{n}, t_{n}^{-1}, \ldots, t_{1}^{-1})$ for $t_{1}, \ldots, t_{n} \in \mathbb{C}^{\times}$. Let $\mathcal{B} \supset \mathcal{T}$ be the Borel subgroup of $\widehat{\mathbf{G}}$ consisting of upper triangular matrices in $\widehat{\mathbf{G}}$. The corresponding set of simple roots is $\Delta^{*} = \{e_{1}^{*} - e_{2}^{*}, \ldots, e_{n-1}^{*} - e_{n}^{*}, e_{n-1}^{*} + e_{n}^{*}\}$ where $e_{j}^{*}(t) = t_{j}$. There exists a pinning $(X_{\alpha^{*}})_{\alpha^{*} \in \Delta^{*}}$ for $(\mathcal{B}, \mathcal{T})$ preserved by $\{1\} \rtimes \operatorname{Gal}(\mathbb{C}/\mathbb{R}) \subset {}^{L}\mathbf{G}$. Thanks to the choice of Borel subgroups there is a canonical isomorphism $\widehat{\mathbf{T}^{*}} \simeq \mathcal{T}$, and we chose the notation for Δ^{*} accordingly. The class of z in $H^{1}(\mathbb{R}, \mathbf{T}^{*})$ corresponds to the character $e_{2}^{*} + e_{4}^{*} + \cdots + e_{2|n/2|}^{*} \in X^{*}(\mathcal{T})/2X^{*}(\mathcal{T}) = X_{*}(\mathbf{T}^{*})/2X_{*}(\mathbf{T}^{*})$ of $\{t \in \mathcal{T} \mid t^{2} = 1\}$.

Let λ be a dominant weight for \mathbf{G}^* , i.e. $\lambda = \lambda_1 e_1 + \cdots + \lambda_n e_n$ where $\lambda_1 \geq \cdots \geq \lambda_{n-1} \geq |\lambda_n|$ are integers. The corresponding discrete parameter φ_{λ} can be chosen (in the canonical $\widehat{\mathbf{G}}$ -conjugacy class) so that for all $z \in W_{\mathbb{C}}$, $\varphi_{\lambda}(z) = \operatorname{diag}((z/\overline{z})^{\lambda_1+n-1}, \ldots, (z/\overline{z})^{\lambda_n}, \ldots) \in \mathcal{T}$, i.e. $\varphi_{\lambda}|_{W_{\mathbb{C}}}$ is dominant for $(\mathcal{B}, \mathcal{T})$. Thus the unique element of the L-packet $\Pi_{\varphi_{\lambda}}(\mathbf{G})$ corresponds to the character $e_2^* + e_4^* + \cdots + e_{2\lfloor n/2 \rfloor}^*$ of $\{t \in \mathcal{T} \mid t^2 = 1\} = C_{\varphi_{\lambda}}$. If n is even and if we choose the other Whittaker datum, the associated character becomes $e_1^* + \cdots + e_{n-1}^*$.

3.2.2 Non-tempered parameters: Adams-Johnson packets

We now turn to the non-tempered case. As before \mathbf{G} will denote any inner form of \mathbf{G}^* . Let $\psi : W_{\mathbb{R}} \times \mathrm{SL}_2(\mathbb{C}) \to {}^L \mathbf{G}$ be an Arthur-Langlands parameter which is bounded on $W_{\mathbb{R}}$. We say that ψ is an Adams-Johnson parameter if the infinitesimal character of ψ is algebraic regular. This is the case if ψ is trivial on $\mathrm{SL}_2(\mathbb{C})$ and defines a discrete Langlands parameter, but there are other examples as well. An Adams-Johnson parameter is always discrete, i.e. its centraliser in $\widehat{f G}$ is contained in $Z(\widehat{f G})$, but not all discrete Arthur-Langlands parameter are Adams-Johnson parameters. From now on we assume that ψ is an Adams-Johnson parameter. We refer to [Taïa, $\S4.2.2$] for the details of what follows. Recall that there is a Langlands parameter φ_{ψ} associated with ψ , and that this parameter is not tempered if $\psi|_{\mathrm{SL}_2(\mathbb{C})}$ is not trivial. Up to conjugating by $\widehat{\mathbf{G}}$ we can assume that $\varphi_{\psi}|_{W_{\mathbb{C}}}$ takes values in \mathcal{T} and that the holomorphic part of $\varphi_{\psi}|_{W_{\mathbb{C}}}$ coincides with that of $\varphi_{\lambda}|_{W_{\mathbb{C}}}$ as above, for a dominant weight λ for **G**. Then, defining C_{ψ} , C_{ψ}^+ and \mathcal{S}_{ψ}^+ as in the tempered case, we have $C_{\psi} \subset C_{\varphi_{\lambda}}, C_{\psi}^+ \subset C_{\varphi_{\lambda}}^+$ and $\mathcal{S}_{\psi}^+ \subset \mathcal{S}_{\varphi_{\lambda}}^+$. Adams and Johnson [AJ87] have attached a packet $\Pi^{AJ}_{\psi}(\mathbf{G})$ of irreducible representations of $\mathbf{G}(\mathbb{R})$ to ψ , and proved endoscopic character relations similar to (2), but only up to a multiplicative constant because only *relative* transfer factors were known at the time. As in [Taïa, §4.2.2] which only adressed the quasi-split case, we will argue that knowing the tempered case above is enough to remove this ambiguity. Let $\mathcal{L} = \operatorname{Cent}(\psi(W_{\mathbb{C}}), \widehat{\mathbf{G}})$. Similarly to the tempered case, $\Pi_{\psi}^{AJ}(\mathbf{G})$ is in bijection with a certain set $\Sigma_{\mathcal{L}}(\mathbf{G})$ of $\mathbf{G}(\mathbb{R})$ -conjugacy classes of pairs (\mathbf{Q}, \mathbf{L}) , and there is a natural surjection $\Sigma_{\mathcal{B}}(\mathbf{G}) \to \Sigma_{\mathcal{L}}(\mathbf{G})$. The choice of a Whittaker datum gives us a distinguished element $cl(\mathbf{Q}^*, \mathbf{L}^*) \in \Sigma_{\mathcal{L}}(\mathbf{G}^*)$, that is the image of $cl(\mathbf{B}^*, \mathbf{T}^*) \in \Sigma_{\mathcal{B}}(\mathbf{G}^*)$. There is a natural bijection between $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{L}^*)$ and the set of isomorphism classes of pairs $((\mathbf{G}, \Xi, z), (\mathbf{Q}, \mathbf{L}))$ where (\mathbf{G}, Ξ, z) is a rigid inner twist of \mathbf{G}^* and $\operatorname{cl}(\mathbf{Q}, \mathbf{L}) \in \Sigma_{\mathcal{L}}(\mathbf{G})$. This realises each such \mathbf{L} as a rigid inner twist of \mathbf{L}^* . Now fix a rigid inner twist (\mathbf{G}, Ξ, z) of \mathbf{G}^* . By [Art89, §5], if $cl(\mathbf{B}_1, \mathbf{T}_1), cl(\mathbf{B}_2, \mathbf{T}_2) \in \Sigma_{\mathcal{B}}(\mathbf{G})$ map to the same element in $\Sigma_{\mathcal{L}}(\mathbf{G})$, then their associated characters of $\mathcal{S}^+_{\varphi_{\lambda}}$ coincide on \mathcal{S}^+_{ψ} . One can define these characters more directly as follows. Observe that the preimage $\widetilde{\mathcal{L}}$ of \mathcal{L} in $\widehat{\overline{\mathbf{G}}}$ is isomorphic to $\widehat{\overline{\mathbf{L}}} = \widehat{\overline{\mathbf{L}^*}}$, and that the $\operatorname{Gal}(\mathbb{C}/\mathbb{R})$ actions coincide if we let $\operatorname{Gal}(\mathbb{C}/\mathbb{R})$ act on $\widetilde{\mathcal{L}}$ via $\psi|_{W_{\mathbb{R}}}$. The pairing between $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{L}^*)$ and $\mathcal{S}^+_{\psi} \simeq Z(\widehat{\overline{\mathbf{L}}})^+$ is naturally the one given by [Kalb, Corollary 5.4]. Thus we get a mapping $\Pi_{\psi}^{AJ}(\mathbf{G}) \rightarrow$ $\left(\mathcal{S}_{\psi}^{+}\right)^{\vee}$, which as in the tempered case we simply denote by $\pi \mapsto \langle \cdot, \pi \rangle$. As in the tempered case it depends on (Ξ, z) and \mathfrak{w} . Again the restriction of each $\pi \mapsto \langle \cdot, \pi \rangle$ to $Z(\widehat{\overline{\mathbf{G}}})^+$ is the character associated to the class of z in $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{G}^*)$. Note that in general the mapping $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{L}^*) \to \left(\mathcal{S}_{\psi}^+\right)^{\vee}$ is not injective: it factors through the group $H^1_{\mathrm{sc}}(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{L}^*)$ defined in [Kalb, §3.4].

Remark 3.2.4. As in the tempered case and Remark 3.2.2, in the case of an anisotropic rigid inner twist (\mathbf{G}, Ξ, z) , for any Adams-Johnson parameter ψ the packet $\Pi_{\psi}^{AJ}(\mathbf{G})$ is a singleton.

Observe that there is a canonical lift of $s_{\psi} \in C_{\psi}$ to C_{ψ}^+ , since $SL_2(\mathbb{C})$ is simply con-

nected. In fact s_{ψ} is well-defined in $Z(\mathcal{L}_{sc})^{\operatorname{Gal}(\mathbb{C}/\mathbb{R})} = Z(\widehat{\mathbf{L}_{ad}})^{\operatorname{Gal}(\mathbb{C}/\mathbb{R})}$: $\psi|_{\operatorname{SL}_2(\mathbb{C})} : \operatorname{SL}_2(\mathbb{C}) \to \mathcal{L}$ is the principal morphism and thus s_{ψ} is equal to the sum of the positive coroots for \mathcal{L} evaluated at -1, for any choice of ordering on the coroots. We still denote its image in C_{ψ}^+ and \mathcal{S}_{ψ}^+ by s_{ψ} . We can now prove the precise endoscopic character relations, refining [AJ87, Theorem 2.21].

Proposition 3.2.5. Let $\dot{s} \in S_{\psi}^+$, which induces a refined endoscopic datum $\dot{\mathbf{e}} = (\mathbf{H}, \mathcal{H}, \dot{s}, \xi)$ for \mathbf{G} , and choose an L-embedding ${}^L\xi : {}^L\mathbf{H} \to {}^L\mathbf{G}$. Denote by ψ' the Adams-Johnson parameter for \mathbf{H} such that $\psi = {}^L\xi \circ \psi'$. For any $\Delta'[\dot{\mathbf{e}}, {}^L\xi, \mathbf{w}, \Xi, z]$ -matching smooth compactly supported distributions f(g)dg on $\mathbf{G}(\mathbb{R})$ and f'(h)dh on $\mathbf{H}(\mathbb{R})$, we have

$$e(\mathbf{G})\sum_{\pi\in\Pi^{\mathrm{AJ}}_{\psi}(\mathbf{G})}\langle\dot{s}s_{\psi},\pi\rangle\mathrm{tr}\left(\pi(f(g)dg)\right) = \sum_{\pi'\in\Pi^{\mathrm{AJ}}_{\psi'}(\mathbf{H})}\langle s_{\psi'},\pi'\rangle\mathrm{tr}\left(\pi'(f'(h)dh)\right).$$
 (4)

Proof. Denote by $\pi_{\psi,\mathbf{Q},\mathbf{L}} \in \Pi^{\mathrm{AJ}}_{\psi}(\mathbf{G})$ the representations associated with $\mathrm{cl}(\mathbf{Q},\mathbf{L}) \in \Sigma_{\mathbf{L}}(\mathbf{G})$. As usual denote $q(\mathbf{L}) = \frac{1}{2} \dim \mathbf{L}(\mathbb{R})/K_{\mathbf{L}}$ for any maximal compact subgroup $K_{\mathbf{L}}$ of $\mathbf{L}(\mathbb{R})$. Recall Kottwitz' sign $e(\mathbf{L}) = (-1)^{q(\mathbf{L})-q(\mathbf{L}_{qs})}$ where \mathbf{L}_{qs} is the quasi-split inner form of \mathbf{L} . We need a preliminary result similar to [Kot90, Lemma 9.1] or [Art89, Lemma 5.1], namely that $\langle s_{\psi}, \pi \rangle = e(\mathbf{L})$. By [Taïa, Lemma 4.2.1] the group \mathbf{L}^* is quasi-split, i.e. $\mathbf{L}_{qs} = \mathbf{L}^*$. The representation π corresponds to an element of $H^1(u_{\mathbb{R}} \to \mathcal{E}_{\mathbb{R}}, \mathbf{Z} \to \mathbf{L}^*)$, and $\langle s_{\psi}, \pi \rangle$ is obtained using the pairing [Kalb]. This pairing is compatible with the pairing between $H^1(\mathbb{R}, \mathbf{L}^*_{ad})$ and $Z(\widehat{\mathbf{L}}_{sc})^{\mathrm{Gal}(\mathbb{C}/\mathbb{R})}$ and s_{ψ} is well-defined in the latter, as the sum of the positive coroots evaluated at -1. It is well-known that this recovers $e(\mathbf{L})$.

Reformulating [AJ87, Theorem 2.21] using this preliminary result, we get that there exists $c \in \mathbb{C}^{\times}$ such that

$$e(\mathbf{G})\sum_{\pi\in\Pi^{\mathrm{AJ}}_{\psi}(\mathbf{G})}\langle\dot{s}s_{\psi},\pi\rangle\mathrm{tr}\left(\pi(f(g)dg)\right)=c\sum_{\pi'\in\Pi^{\mathrm{AJ}}_{\psi'}(\mathbf{H})}\langle s_{\psi'},\pi'\rangle\mathrm{tr}\left(\pi'(f'(h)dh)\right),$$

and we are left to prove that c = 1. We proceed as in the quasi-split case [Taïa, §4.2.2]: we restrict to distributions f(g)dg whose support is contained in semisimple elliptic regular conjugacy classes of $\mathbf{G}(\mathbb{R})$. Let $\varphi = \varphi_{\lambda}$ be the discrete Langlands parameter aligned with ψ as above. Then using Johnson's resolution of Adams-Johnson representations [Joh84],

$$\sum_{\pi \in \Pi_{\psi}^{\mathrm{AJ}}(\mathbf{G})} \langle \dot{s}s_{\psi}, \pi \rangle \mathrm{tr} \left(\pi(f(g)dg) \right) = (-1)^{q(\mathbf{L}^*)} \sum_{\pi \in \Pi_{\varphi}(\mathbf{G})} \langle \dot{s}, \pi \rangle \mathrm{tr} \left(\pi(f(g)dg) \right)$$

and similarly for **H**. Note that the sign $(-1)^{q(\mathbf{L}^*)}$ is identical for ψ and ψ' , since the groups $\mathcal{L} \subset \widehat{\mathbf{G}}$ and $\mathcal{L}' \subset \widehat{\mathbf{H}}$ are naturally isomorphic, compatibly with their Galois actions. As a consequence of (3) and these equalities for ψ and ψ' , the relation (4) holds for such distributions f(g)dg. There exists such a distribution for which the left hand side of (4) does not vanish, and thus c = 1 and (4) is valid for any distribution.

3.2.3 Relation with Arthur packets

For our global purposes, which will rely on [Art13], we would like to replace $\Pi_{\psi'}^{AJ}(\mathbf{H})$ in the right hand side of equation (4) by Arthur's packet $\Pi_{\psi'}(\mathbf{H})$ reviewed in section 2.5, that is

$$e(\mathbf{G})\sum_{\pi\in\Pi^{\mathrm{AJ}}_{\psi}(\mathbf{G})}\langle \dot{s}s_{\psi},\pi\rangle\mathrm{tr}\left(\pi(f(g)dg)\right) = \sum_{\pi'\in\Pi_{\psi'}(\mathbf{H})}\langle s_{\psi'},\pi'\rangle\mathrm{tr}\left(\pi'(f'(h)dh)\right).$$
 (5)

For this to even make sense it is necessary for all the stable orbital integrals of f'(h)dh to be invariant under the subgroup $Out(\mathbf{H}_1) \times Out(\mathbf{H}_2)$ of $Out(\mathbf{H})$, where $\mathbf{H} \simeq \mathbf{H}_1 \times \mathbf{H}_2$ as in section 2.3.

In the quasi-split case, more precisely for the trivial rigid inner twist of \mathbf{G}^* , it is known (and was used by Arthur) that this is the case if f(g)dg has $\operatorname{Out}(\mathbf{G})$ -invariant orbital integrals, for the choice of rational outer automorphism described in section 2.5. In this case Arancibia, Moeglin and Renard [AMR] recently proved that for any Adams-Johnson parameter ψ , the packets of Adams-Johnson and Arthur and the maps to $(\mathcal{S}_{\psi})^{\vee}$ coincide: this is equivalent to equation (5) for all ψ but with $\dot{s} = 1$, by Fourier inversion on \mathcal{S}_{ψ} .

Consequently for a non-trivial rigid inner twist (\mathbf{G}, Ξ, z) equation (5) holds provided that all stable orbital integrals of f'(h)dh are invariant under $\operatorname{Out}(\mathbf{H}_1) \times \operatorname{Out}(\mathbf{H}_2)$. As explained in Remark 3.1.1, in the even orthogonal case to even formulate the $\operatorname{Out}(\mathbf{G})$ invariance of orbital integrals of f(g)dg, we have to restrict to groups \mathbf{G} realised as *pure* inner twists (\mathbf{G}, Ξ, z) , i.e. assume that $z \in Z^1(\mathbb{R}, \mathbf{G}^*)$. We make this assumption and treat the three types of groups separately. As before we denote by $\dot{\mathbf{c}}$ the refined endoscopic datum $(\mathbf{H}, \mathcal{H}, \dot{s}, \xi)$ and by \mathbf{c} the underlying endoscopic datum, and fix an L-embedding ${}^L\xi: {}^L\mathbf{H} \to {}^L\mathbf{G}$.

- If $\mathbf{G}^* = \mathbf{SO}_{2n+1}$, the group $\operatorname{Out}(\mathbf{H}_1) \times \operatorname{Out}(\mathbf{H}_2)$ is always trivial and so there is nothing to prove.
- If $\mathbf{G}^* = \mathbf{Sp}_{2n}$ and $\mathbf{H} \simeq \mathbf{Sp}_{2a} \times \mathbf{SO}_{2b}^{\alpha}$ with b > 0, then $\operatorname{Out}(\mathbf{H}_1)$ is trivial, $\operatorname{Out}(\mathbf{H}_2) \simeq \mathbb{Z}/2\mathbb{Z}$ and

$$\operatorname{Out}(\mathbf{H}_1) \times \operatorname{Out}(\mathbf{H}_2) = \operatorname{Out}(\mathbf{H}) \simeq \operatorname{Out}(\mathfrak{e}) = \operatorname{Out}(\mathfrak{e}).$$

The last equality is crucial: in general we only have $\operatorname{Out}(\dot{\mathfrak{e}}) \subset \operatorname{Out}(\mathfrak{e})$, while the transfer factors defined in [Kalb, §5.3] are only invariant by isomorphisms of *re-fined* endoscopic data [Kalb, Proposition 5.6]. Furthermore, the L-embedding ${}^{L}\xi$ is preserved by $\operatorname{Out}(\dot{\mathfrak{e}})$. Therefore, if $\operatorname{cl}(\gamma_{1})$ and $\operatorname{cl}(\gamma_{2})$ are stable conjugacy classes of semisimple regular elements of $\mathbf{H}(\mathbb{R})$ exchanged by the outer automorphism of \mathbf{H} , then γ_{1} is a norm of a conjugacy class $\operatorname{cl}(\delta)$ in $\mathbf{G}(\mathbb{R})$ (in the sense of [KS99, §3]) if and only if γ_{2} is a norm of $\operatorname{cl}(\delta)$, and if this is the case then

$$\Delta'[\dot{\mathfrak{e}}, {}^{L}\xi, \mathfrak{w}, \Xi, z](\gamma_1, \delta) = \Delta'[\dot{\mathfrak{e}}, {}^{L}\xi, \mathfrak{w}, \Xi, z](\gamma_2, \delta).$$

This implies that for any smooth compactly supported distribution f(g)dg on $\mathbf{G}(\mathbb{R})$, the stable orbital integrals of any smooth compactly supported distribution f'(h)dhon $\mathbf{H}(\mathbb{R})$ which is a transfer of f(g)dg are invariant under $\operatorname{Out}(\mathbf{H})$. • The last case to consider is $\mathbf{G}^* = \mathbf{SO}_{2n}^{\alpha}$ and $\mathbf{H} \simeq \mathbf{SO}_{2a}^{\beta} \times \mathbf{SO}_{2b}^{\gamma}$. Using the same arguments as in the previous case, one could show that if a, b > 0 then the transfer factors are invariant under (θ_1, θ_2) where θ_i is the non-trivial outer automorphism of \mathbf{H}_i . We still have to consider the remaining coset in $Out(\mathbf{H}_1) \times Out(\mathbf{H}_2)$, in parallel with the non-trivial outer automorphism θ of **G**. Since we only consider groups **G** realised as *pure* inner twists (\mathbf{G}, Ξ, z) we can use Waldspurger's explicit formulae [Wal10] for the transfer factors. One can lift θ to an automorphism of **G** defined over \mathbb{R} in a natural way since **G** is naturally a special orthogonal group, by considering any \mathbb{R} -point of the non-trivial connected component of the corresponding orthogonal group. We still denote by θ any such lift, it is unique up to composing it with $\operatorname{Ad}(q)$ for some $q \in \mathbf{G}(\mathbb{R})$. One can simply observe on Waldspurger's formulae that the transfer factors are invariant under the action of (θ_1, θ_2) on **H** if a, b > 0, as well as under the simultaneous action of θ on **G** and the non-trivial element of $(Out(\mathbf{H}_1) \times Out(\mathbf{H}_2)) / \{1, (\theta_1, \theta_2)\}$ (resp. $Out(\mathbf{H})$) if a, b > 0 (resp. if ab = 0). To be precise we ought to remark that Waldspurger leaves out the ϵ -factor of [KS99, $\{5.3\}$, but this is just a constant that does not change the invariance properties, and that Waldspurger computed Δ whereas we use Δ' in this paper. This is not a real issue since the factors Δ' , defined in [KS] and [Kalb], are equal to the factors Δ for the refined endoscopic datum obtained by replacing \dot{s} by \dot{s}^{-1} .

This implies that for any smooth compactly supported distribution f(g)dg on $\mathbf{G}(\mathbb{R})$ having θ -invariant orbital integrals, the stable orbital integrals of any smooth compactly supported distribution f'(h)dh on $\mathbf{H}(\mathbb{R})$ which is a transfer of f(g)dg are invariant under $\operatorname{Out}(\mathbf{H}_1) \times \operatorname{Out}(\mathbf{H}_2)$.

For clarity, let us restate what we have deduced from [AMR].

Proposition 3.2.6. Let (\mathbf{G}, Ξ, z) be a rigid inner twist of a reductive group \mathbf{G}^* over \mathbb{R} as in section 2.1. Assume that \mathbf{G}^* admits discrete series. If \mathbf{G}^* is even orthogonal, assume also that $z \in Z^1(\mathbb{R}, \mathbf{G}^*)$ and let θ be an automorphism of \mathbf{G} as in the above discussion. Let ψ be an Adams-Johnson parameter for \mathbf{G} . Let $\dot{\mathbf{s}} \in C^+_{\psi}$, let $\mathbf{c} = (\mathbf{H}, \mathcal{H}, \dot{\mathbf{s}}, \xi)$ be the induced elliptic refined endoscopic datum, and choose an L-embedding ^L ξ for this endoscopic datum. As explained in section 2.3 \mathbf{H} decomposes as a product $\mathbf{H}_1 \times \mathbf{H}_2$. Let f(g)dg be a smooth compactly supported distribution on $\mathbf{G}(\mathbb{R})$. In the case $\mathbf{G}^* = \mathbf{SO}_{2n}^{\alpha}$ we assume that the orbital integrals of f(g)dg are θ -invariant. Let f'(h)dh be a transfer of f(g)dg with respect to the transfer factors $\Delta'[\dot{\mathbf{c}}, {}^L\xi, \mathbf{w}, \Xi, z]$.

Then the stable orbital integrals of f'(h)dh are invariant under $Out(\mathbf{H}_1) \times Out(\mathbf{H}_2)$ and the endoscopic character relation (5) holds.

3.3 Trivial inner twists

In this section we denote by F a local field of characteristic zero, and consider a quasi-split reductive group \mathbf{G}^* over F as in section 2.1, and a finite subgroup \mathbf{Z} of the center of \mathbf{G}^* . Consider also a rigid inner twist (\mathbf{G}, Ξ, z) of \mathbf{G}^* . Denote by $\overline{\mathbf{G}}^*$ and $\overline{\mathbf{G}}$ the quotients of \mathbf{G}^* and \mathbf{G} by \mathbf{Z} . Assume that the image of z in $H^1(F, \mathbf{G}^*_{ad})$ is trivial. For example this is what we obtain by localisation at a place $v \notin S$ of a global rigid inner twist as in section 3.1.2. Fix a Whittaker datum \boldsymbol{w} for \mathbf{G}^* .

If the class of z in $H^1(u_F \to \mathcal{E}_F, \mathbb{Z} \to \mathbb{G}^*)$ is trivial, then (\mathbb{G}, Ξ, z) is isomorphic to the trivial rigid inner twist $(\mathbb{G}^*, \mathrm{Id}, 1)$ and the isomorphism is unique up to $\mathbb{G}^*(F)$. In particular \mathbb{G} inherits a well-defined Whittaker datum \mathfrak{w}' from \mathfrak{w} , and the absolute transfer factors $\Delta'[\dot{\mathfrak{e}}, {}^L\xi, \mathfrak{w}, \Xi, z]$ are equal to the absolute transfer factors $\Delta'[\mathfrak{e}, {}^L\xi, \mathfrak{w}']$ defined in [KS99, §5.3]. Therefore we can simply use the Arthur packets reviewed in section 2.5 for \mathbb{G} and \mathfrak{w}' .

If however the class of z in $H^1(u_F \to \mathcal{E}_F, \mathbb{Z} \to \mathbb{G}^*)$ is non-trivial, as can happen by Remark 3.1.3, then an isomorphism between \mathbb{G} and \mathbb{G}^* is only well-defined up to $\mathbb{G}^*_{\mathrm{ad}}(F)$, which does not preserve the $\mathbb{G}^*(F)$ -orbit of \mathfrak{w} in general. Thus there is no natural choice of Whittaker datum for \mathbb{G} , and even if there was the corresponding absolute transfer factors would have no reason to coincide with $\Delta'[\dot{\mathfrak{e}}, {}^L\xi, \mathfrak{w}, \Xi, z]$. We conclude that in this situation we cannot merely use the packets reviewed in section 2.5 along with the characters $\langle \cdot, \pi \rangle$ to get endoscopic character relations similar to equation (2) but for the transfer factors $\Delta'[\dot{\mathfrak{e}}, {}^L\xi, \mathfrak{w}, \Xi, z]$.

Kaletha [Kalc, §6.3] computed the effect of twisting the cocycle $z \in Z^1(u_F \to \mathcal{E}_F, \mathbf{Z} \to \mathbf{G}^*)$ by an element of $Z^1(u_F \to \mathcal{E}_F, \mathbf{Z} \to \mathbf{Z}(\mathbf{G}^*))$ on the transfer factors $\Delta'[\dot{\mathbf{e}}, {}^L\xi, \mathbf{w}, \Xi, z]$ [Kalc, Lemma 6.3], and deduced the effect on the spectral transfer factors $\langle \dot{s}, \pi \rangle$ [Kalc, Lemma 6.2]. Let us recall how this applies to the present situation. Up to replacing (\mathbf{G}, Ξ, z) by an isomorphic rigid inner twist, we can assume that the image of z in $Z^1(F, \mathbf{G}^*_{\mathrm{ad}})$ is trivial, i.e. that $z \in Z^1(u_F \to \mathcal{E}_F, \mathbf{Z} \to \mathbf{Z}(\mathbf{G}^*))$. In particular, the isomorphism $\Xi : \mathbf{G}^*_F \to \mathbf{G}_F$ is defined over F. Up to enlarging \mathbf{Z} (as a finite subgroup of $\mathbf{Z}(\mathbf{G}^*)$), we can assume that $z \in Z^1(u_F \to \mathcal{E}_F, \mathbf{Z} \to \mathbf{Z})$. The exact sequence $1 \to \mathbf{Z} \to \mathbf{G}^* \to \overline{\mathbf{G}}^* \to 1$ dualises as $1 \to \widehat{\mathbf{Z}} \to \widehat{\mathbf{G}} \to \widehat{\mathbf{G}} \to 1$, where $\widehat{\mathbf{Z}}$ is naturally identified with $2i\pi\mathbb{Z}\otimes X^*(\mathbf{Z})$. Let $\psi: \mathrm{WD}_F \times \mathrm{SL}_2(\mathbb{C}) \to {}^L\mathbf{G}$ be an Arthur-Langlands parameter. As in the real case, let $C_{\psi} = \mathrm{Cent}(\psi, \widehat{\mathbf{G}}), C_{\psi}^+$ its preimage in $\widehat{\mathbf{G}}$ and $\mathcal{S}^+_{\psi} = \pi_0(C_{\psi}^+)$. Kaletha constructs a morphism $d: \mathcal{S}^+_{\psi} \to Z^1(\mathrm{Gal}_F, \widehat{\mathbf{Z}})$. By [Kalc, §6.2] there is a perfect pairing

$$Z^{1}(\operatorname{Gal}_{F}, \widehat{\mathbf{Z}}) \times H^{1}(u_{F} \to \mathcal{E}_{F}, \mathbf{Z} \to \mathbf{Z}) \to \mathbb{C}^{\times}$$
 (6)

which we simply denote by $\langle \cdot, \cdot \rangle_{\mathbf{Z}}$. These are the two constructions needed to state the existence of Arthur packets for **G**. To avoid confusion we will denote by $\Pi_{\psi}(\mathbf{G}^*)$ and θ^* what was denoted by Π_{ψ} and θ in section 2.5. If $\mathbf{G}^* = \mathbf{SO}_{2n}^{\alpha}$, let $\theta = \Xi \circ \theta^* \circ \Xi^{-1}$. Define a multiset $\Pi_{\psi}(\mathbf{G})$ of representations of $\mathbf{G}(F)$, or θ -orbits of representations in the even orthogonal case, as $(\Xi^{-1})^*(\Pi_{\psi}(\mathbf{G}^*))$, and for $\pi \in \Pi_{\psi}(\mathbf{G})$ define a character $\langle \cdot, \pi \rangle_{\mathbf{G}}$ of \mathcal{S}^+_{ψ} by

$$\langle \dot{s}, \pi \rangle = \langle d(\dot{s})^{-1}, z \rangle_{\mathbf{Z}} \times \langle s, \pi \circ \Xi \rangle$$

for $\dot{s} \in \mathcal{S}_{\psi}^+$ and s its image in \mathcal{S}_{ψ} .

Fix $\dot{s} \in C_{\psi}^+$, let $\mathfrak{e} = (\mathbf{H}, \mathcal{H}, \dot{s}, \xi)$ be the induced refined endoscopic datum, and choose an L-embedding ${}^L\xi : {}^L\mathbf{H} \to {}^L\mathbf{G}$ for this endoscopic datum. As explained in section 2.3, **H** decomposes as a product $\mathbf{H}_1 \times \mathbf{H}_2$. **Proposition 3.3.1.** Let f(g)dg be a smooth compactly supported distribution on $\mathbf{G}(F)$. In the even orthogonal case assume that the orbital integrals of f(g)dg are invariant under θ . Let f'(h)dh be a transfer of f(g)dg with respect to the transfer factors $\Delta'[\dot{\mathbf{e}}, {}^L\xi, \mathbf{w}, \Xi, z]$.

Then the stable orbital integrals of f'(h)dh are invariant under $Out(\mathbf{H}_1) \times Out(\mathbf{H}_2)$ and we have

$$\sum_{\pi \in \Pi_{\psi}(\mathbf{G})} \langle \dot{s}s_{\psi}, \pi \rangle \operatorname{tr}\left(\pi(f(g)dg)\right) = \sum_{\pi' \in \Pi_{\psi'}(\mathbf{H})} \langle s_{\psi'}, \pi' \rangle \operatorname{tr}\left(\pi'(f'(h)dh)\right)$$
(7)

Proof. This is a special case of [Kalc, Lemma 6.2], except for the invariance under $\operatorname{Out}(\mathbf{H}_1) \times \operatorname{Out}(\mathbf{H}_2)$ of the stable orbital integrals of f'(h)dh. The corresponding invariance property for the transfer factors $\Delta'[\dot{\mathbf{e}}, {}^L\xi, \mathfrak{w}, \Xi, z]$ can be proved as in Proposition 3.2.6. More simply, it can be deduced from the similar invariance property for the transfer factors for \mathbf{G}^* , since it is a property of the *relative* transfer factors.

Remark 3.3.2. By Remark 3.1.3 in the global setting of section 3.1.2 the only case where we cannot reduce to z = 1 at any finite place is when $\mathbf{G}^* = \mathbf{SO}_{2n}^{\alpha}$. In this case the restriction of $z \in Z^1(u_F \to \mathcal{E}_F, \mathbf{Z} \to \mathbf{Z})$ to u_F is trivial, that is $z \in Z^1(F, \mathbf{Z})$. Instead of the pairing (6) of [Kalc, §§6.1-2] one can use the usual Poitou-Tate pairing

$$H^1(\operatorname{Gal}_F, \widehat{\mathbf{Z}}) \times H^1(F, \mathbf{Z}) \to \mathbb{C}^{\times}$$

and the characters $\langle \dot{s}, \pi \rangle$ defined above for $\pi \in \Pi_{\psi}(\mathbf{G})$ descend to characters of $\pi_0(C_{\psi})$, which is always a 2-group, in particular it is abelian.

3.4 Adèlic Arthur packets for rigid inner forms

Let us go back to the setting of section 3.1.2: F is a totally real number field, \mathbf{G}^* one of the quasi-split reductive groups over F defined in section 2.1. Let \mathbf{Z} be the center of \mathbf{G}^* if \mathbf{G}^* is symplectic. If \mathbf{G}^* is orthogonal we let $\mathbf{Z} = 1$. Let S be a non-empty set of real places of F, and consider a rigid inner twist (\mathbf{G}, Ξ, z) of \mathbf{G}^* satisfying property \star with respect to S, where $z \in Z^1(P_{F,\dot{V}} \to \mathcal{E}_{F,\dot{V}}, \mathbf{Z} \to \mathbf{G}^*)$. Such rigid inner twists were classified in Proposition 3.1.2. Note that if \mathbf{G}^* is orthogonal z is simply an element of $Z^1(F, \mathbf{G}^*)$. Fix also a global Whittaker datum \mathfrak{w} for \mathbf{G}^* .

For any place v of F we get a rigid inner twist $(\mathbf{G}_{F_v}, \Xi_v, z_v)$ of $\mathbf{G}_{F_v}^*$, where

$$z_v \in \begin{cases} Z^1(F_v, \mathbf{G}_{F_v}^*) & \text{if } \mathbf{G}^* \text{ is orthogonal or } v \notin S, \\ Z^1(u_v \to \mathcal{E}_{F_v}, \mathbf{Z} \to \mathbf{G}_{F_v}^*) & \text{if } \mathbf{G}^* \text{ is symplectic and } v \in S \end{cases}$$

is the localisation of z at v and $\Xi_v = \Xi_{\overline{F}_v}$. We also get a Whittaker datum \mathfrak{w}_v for $\mathbf{G}_{F_v}^*$.

Recall that \mathbf{G}^* (resp. \mathbf{G}) admits a model over $\mathcal{O}_F[\frac{1}{N}]$ for some $N \geq 1$, and that up to replacing N by a multiple two such models are canonically isomorphic. Furthermore, there exists N such that Ξ is defined over a finite étale extension of $\mathcal{O}_F[\frac{1}{N}]$. For almost all places v, z_v is the inflation of an unramified cocycle, and since $H^1(\mathcal{O}_{F_v}, \mathbf{G}^*)$ is trivial we get an isomorphism Ξ'_v between the models of \mathbf{G}^* and \mathbf{G} over \mathcal{O}_{F_v} , well-defined up to the image of $\mathbf{G}^*(\mathcal{O}_{F_v})$ in $\mathbf{G}^*_{\mathrm{ad}}(\mathcal{O}_{F_v})$. Finally, at almost all places v the Whittaker datum \mathbf{w}_v is compatible with the choice of hyperspecial maximal compact subgroup $\mathbf{G}^*(\mathcal{O}_{F_v})$ of $\mathbf{G}^*(F_v)$ (in the sense of [CS80]), and this compatibility can be pulled back to \mathbf{G} via $(\Xi'_v)^{-1}$. At the finite set of places not in S where v is Archimedean or \mathbf{G} is ramified or z_v is ramified, there exists $g \in \mathbf{G}^*(\overline{F_v})$ such that $\Xi'_v := \operatorname{Ad}(g) \circ \Xi_v$ is defined over F_v , and Ξ'_v is well-defined up to $\mathbf{G}^*_{\operatorname{ad}}(F_v)$. If the image of $z_v \in H^1(F_v, \mathbf{G}^*)$ is trivial we can further impose that $z_v(\sigma) = g\sigma(g)^{-1}$ for all $\sigma \in \operatorname{Gal}_{F_v}$, and Ξ_v is then well-defined up to the image of $\mathbf{G}^*(F_v)$ in $\mathbf{G}^*_{\operatorname{ad}}(F_v)$.

If $\mathbf{G}^* = \mathbf{SO}_{2n}^{\alpha}$, let θ be an outer automorphism of \mathbf{G} that is realised by an element having determinant -1 in the corresponding orthogonal group. Choose an automorphism θ^* of \mathbf{G}^* similarly. Denote by θ_v and θ_v^* the induced automorphisms of \mathbf{G}_{F_v} and $\mathbf{G}_{F_v}^*$. At all places $v \notin S$, the automorphisms θ_v and $\Xi'_v \circ \theta_v^* \circ (\Xi'_v)^{-1}$ of \mathbf{G} are in the same orbit under $\mathbf{G}(F_v)$.

Let $\Psi_{\text{disc}}^{S-\text{AJ}}(\mathbf{G}^*) \subset \Psi_{\text{disc}}(\mathbf{G}^*)$ be the set of $\dot{\psi}$ such that for any $v \in S$, the localisation $\dot{\psi}_v$ is an Adams-Johnson parameter as in section 3.2.2. Let $\dot{\psi} \in \Psi_{\text{disc}}^{S-\text{AJ}}(\mathbf{G}^*)$, To formulate the multiplicity formula in the next section, we need to define a packet $\Pi_{\dot{\psi}}(\mathbf{G})$ of admissible representations of $\mathbf{G}(\mathbb{A}_F)$, along with a mapping $\Pi_{\dot{\psi}}(\mathbf{G}) \to (\mathcal{S}_{\psi})^{\vee}$, using the local Arthur packets at our disposal (Propositions 3.2.6 and 3.3.1). This was done by Kaletha [Kala, Proposition 4.2] in greater generality. However, as our current setting is slightly different from Kaletha's (we did not lift the cocycles to \mathbf{G}_{sc}^* as Kaletha did) and simpler than the general case since the Hasse principle holds for all the groups considered in this paper, it is worth recalling how Kaletha's construction specialises to the cases at hand.

Let $\Pi_{\dot{\psi}}(\mathbf{G})$ be the set of restricted tensor products $\pi = \bigotimes_{v} \pi_{v}$ where for all places v of F, $\pi_{v} \in \Pi_{\dot{\psi}_{v}}(\mathbf{G}(F_{v}))$ if $v \notin S$ and $\pi_{v} \in \Pi_{\dot{\psi}_{v}}^{\mathrm{AJ}}(\mathbf{G})$ if $v \in S$, and for almost all places v the character $\langle \cdot, \pi_{v} \rangle$ is trivial. For any such π let us define a character $\langle \cdot, \pi \rangle$ of $\mathcal{S}_{\dot{\psi}}$. Recall that $\mathcal{S}_{\dot{\psi}} = C_{\dot{\psi}}/Z(\widehat{\mathbf{G}})^{\mathrm{Gal}_{F}}$ and that $C_{\dot{\psi}}^{+}$ denotes the preimage of $C_{\dot{\psi}}$ in $\widehat{\mathbf{G}}$. Thus it is equivalent to define a character on $C_{\dot{\psi}}^{+}$ which is trivial on $Z(\widehat{\mathbf{G}})^{+}$. Note that in general each $\langle \cdot, \pi_{v} \rangle$ is the character of an irreducible representation of the finite group $\mathcal{S}_{\dot{\psi}_{v}}^{+}$, but as we observed in Remark 3.3.2, in our setting it is simply a morphism $\mathcal{S}_{\dot{\psi}_{v}}^{+} \to \mathbb{C}^{\times}$. Its restriction χ_{v} to $\pi_{0}(Z(\widehat{\mathbf{G}})^{+,v})$ is the image of z via [Kalb, Corollary 5.4]. Thanks to the natural morphisms $\mathcal{S}_{\dot{\psi}}^{+} \to \mathcal{S}_{\dot{\psi}_{v}}^{+}$ reviewed in section 2.6 we can define $\langle \cdot, \pi_{v} \rangle = \prod_{v} \langle \cdot, \pi_{v} \rangle |_{\mathcal{S}_{\psi}^{+}}$. By [Kala, Corollary 3.45] we have $\prod_{v} \chi_{v}|_{\pi_{0}\left(Z(\widehat{\mathbf{G}})^{+}\right)} = 1$ and so $\langle \cdot, \pi \rangle$ is trivial on $\pi_{0}\left(Z(\widehat{\mathbf{G}})^{+}\right)$ and descends to a character of $\mathcal{S}_{\dot{\psi}}$.

For $s \in S_{\dot{\psi}}$ let $\mathbf{e} = (\mathbf{H}, \mathcal{H}, s, \xi)$ be the induced elliptic endoscopic datum. Sensu stricto it depends on the choice of a lift of s in $Z(\widehat{\mathbf{G}})C_{\dot{\psi}}$, but multiplication by an element of $Z(\widehat{\mathbf{G}})$ does not change the isomorphism class of \mathbf{e} . Fix a compatible L-embedding ${}^{L}\xi : {}^{L}\mathbf{H} \to {}^{L}\mathbf{G}$ as in section 2.3, yielding $\dot{\psi}' \in \Psi_{\mathbf{G}-\text{disc}}(\mathbf{H})$ such that $\dot{\psi} = {}^{L}\xi \circ \dot{\psi}'$ as reviewed in section 2.4. For any choice of $\dot{s} \in S_{\dot{\psi}}^+$ above s we get local refined endoscopic data $\dot{\mathbf{e}}_v$ above \mathbf{e}_v , and thus absolute transfer factors $\Delta'[\dot{\mathbf{e}}_v, {}^{L}\xi, \mathbf{G}, \Xi_v, z_v]$. By [Kala, Proposition 4.1] the product of these local transfer factors is the *canonical* adélic absolute transfer factor $\Delta'[\mathbf{e}, {}^{L}\xi, \mathbf{G}, \Xi]$. Recall that this transfer factor does not depend on the choices of z and \mathbf{w} , nor on the choice of \dot{s} above s. Let $f(g)dg = \prod_v f_v(g_v)dg_v$ be a smooth, compactly supported distribution on $\mathbf{G}(\mathbb{A}_F)$ such that for any place v, the orbital integrals of $f_v(g_v)dg_v$ are θ_v -invariant. Let $f'(h)dh = \prod_v f'(h_v)dh_v$ be a transfer of f(g)dg to $\mathbf{H} = \mathbf{H}_1 \times \mathbf{H}_2$. Then the stable orbital integrals of f'(h)dh are invariant under $\operatorname{Out}(\mathbf{H}_{1,F_v}) \times \operatorname{Out}(\mathbf{H}_{2,F_v})$ for all places v, and we have the endoscopic character relation

$$\sum_{\pi \in \Pi_{\dot{\psi}}(\mathbf{G})} \langle ss_{\dot{\psi}}, \pi \rangle \operatorname{tr}\left(\pi(f(g)dg)\right) = \sum_{\pi' \in \Pi_{\dot{\psi}'}(\mathbf{H})} \langle s_{\dot{\psi}'}, \pi' \rangle \operatorname{tr}\left(\pi'(f'(h)dh)\right)$$
(8)

where the packet $\Pi_{\dot{\psi}'}(\mathbf{H})$ on the right hand side is the one defined by Arthur in [Art13, (1.5.3)]. This relation is an immediate consequence of Propositions 3.2.6 and 3.3.1 and the well-known product formula $\prod_{v} e(\mathbf{G}_{F_v}) = 1$ for the Kottwitz signs. Note that $e(\mathbf{G}_{F_v}) = 1$ for any finite place v because \mathbf{G}_{F_v} is quasi-split.

4 Arthur's multiplicity formula for certain inner forms

We continue with the definitions and notations of section 3.4. For any Archimedean place vof F, fix a maximal compact subgroup K_v of $\mathbf{G}(F_v)$ and denote by \mathfrak{g}_v the reductive complex Lie algebra $\mathbb{C} \otimes_{\mathbb{R}} \text{Lie} (\operatorname{Res}_{F_v/\mathbb{R}}(\mathbf{G}_{F_v}))$. Let $\mathcal{A}(\mathbf{G})$ denote the space of automorphic forms for \mathbf{G} in the sense of [BJ79], with respect to $\prod_{v|\infty} K_v$. Let $\mathcal{A}^2(\mathbf{G})$ denote the subspace of square-integrable automorphic forms. It decomposes as a countable orthogonal direct sum of admissible irreducible unitary representations $\pi = \otimes'_v \pi_v$ of $\mathbf{G}(\mathbb{A}_F)$:

$$\mathcal{A}^2(\mathbf{G}) \simeq \bigoplus_{\pi} \pi^{\oplus m(\pi)}.$$

Note that for v Archimedean π_v is really an irreducible unitary (\mathfrak{g}_v, K_v) -module, and our use of the term "representation" is somewhat abusive. For any Archimedean place v of Fthe action of the center $\mathcal{Z}(\mathfrak{g}_v)$ of the enveloping algebra of \mathfrak{g}_v on $\mathcal{A}^2(\mathbf{G})$ is semi-simple, since $\mathcal{Z}(\mathfrak{g}_v)$ acts on π_v via the infinitesimal character $\nu_{\pi_v} : \mathcal{Z}(\mathfrak{g}_v) \to \mathbb{C}$. Recall that via the Harish-Chandra isomorphism such characters are in bijection with semi-simple conjugacy classes in the complex Lie algebra $\widehat{\mathfrak{g}_v}$ dual to \mathfrak{g}_v . These correspond naturally to semi-simple conjugacy classes in the Lie algebra of the complex Lie group $\widehat{\mathbf{G}}$ if v is real, and pairs of such conjugacy classes if v is complex. For an infinitesimal character $\nu = (\nu_v)_{v|\infty}$, denote by $\mathcal{A}^2(\mathbf{G},\nu)$ the subspace of $\mathcal{A}^2(\mathbf{G})$ on which $\mathcal{Z}(\mathfrak{g}_v)$ acts by ν_v for all Archimedean v. If \mathbf{G} is even orthogonal, we will also consider the $\prod_{v|\infty} \operatorname{Out}(\mathbf{G}_{F_v})$ -orbit $\tilde{\nu}$ of ν instead, and $\mathcal{A}^2(\mathbf{G},\tilde{\nu})$ will denote the direct sum of the subspaces $\mathcal{A}^2(\mathbf{G},\nu')$ for ν' in the orbit of ν .

Theorem 4.0.1. Let $f(g)dg = \prod_v f_v(g_v)dg_v$ be a smooth, compactly supported distribution on $\mathbf{G}(\mathbb{A}_F)$ such that for any Archimedean place v of F, the function f_v is $bi-K_v$ -finite. If \mathbf{G} is even orthogonal, assume also that for any place v of F the orbital integrals of $f_v(g_v)dg_v$ are θ_v -invariant. Let $\nu = (\nu_v)_{v|\infty}$ be an infinitesimal character as above, and denote by $\tilde{\nu}$ its $\prod_{v|\infty} \operatorname{Out}(\mathbf{G}_{F_v})$ -orbit, which only consists of ν if \mathbf{G} is symplectic or odd orthogonal. Assume that for all v in the non-empty set S, the infinitesimal character ν_v is algebraic regular. Denote by $\Psi_{\operatorname{disc}}(\mathbf{G}, \tilde{\nu})$ the subset of $\Psi_{\operatorname{disc}}(\mathbf{G})$ consisting of those $\dot{\psi}$ such that for all Archimedean v, the infinitesimal character of $\dot{\psi}_v$ belongs to $\tilde{\nu}_v$. Then for all $\dot{\psi} \in \Psi_{\text{disc}}(\mathbf{G}, \tilde{\nu})$ and $v \in S$, $\dot{\psi}_v$ is an Adams-Johnson parameter in the sense of section 3.2.2, and

$$\operatorname{tr}\left(f(g)dg \,\middle|\, \mathcal{A}^{2}(\mathbf{G},\tilde{\nu})\right) = \sum_{\substack{\dot{\psi} \in \Psi_{\operatorname{disc}}(\mathbf{G},\tilde{\nu})\\up \ to \ \widehat{\mathbf{G}}-\operatorname{conj}\\ s.t. \ \langle\cdot,\pi\rangle = \epsilon_{\psi}}} \sum_{\substack{\pi \in \Pi_{\dot{\psi}}(\mathbf{G})\\tr(\pi(f(g)dg))}} \operatorname{tr}(\pi(f(g)dg)) \tag{9}$$

where ϵ_{ψ} is the character of S_{ψ} defined by Arthur in [Art13, (1.5.6)].

Proof. We use the stabilisation of the trace formula for **G** ([Art13, (3.2.3)], proved in [Art02], [Art01], [Art03]. This technique is hardly new: using the stabilisation of the trace formula for this purpose was initiated in [LL79] for the group SL_2 and its inner forms. We benefited from reading Kottwitz' conjectural but very general stabilisation of the cuspidal tempered part of the spectral side of the trace formula [Kot84], and of course Arthur's book [Art13].

We refer the reader to [Art88, §4] and [Art13, §3.1] for details concerning the spectral side of the trace formula. For each Archimedean place v of F, choose a Cartan subalgebra $\widehat{\mathfrak{h}_v}$ of the complex Lie algebra $\widehat{\mathfrak{g}_v}$ dual to \mathfrak{g}_v . Note that $\widehat{\mathfrak{h}_v}$ has a natural integral structure since it is the Lie algebra of a maximal torus in the dual group of $\operatorname{Res}_{F_v/\mathbb{R}}(\mathbf{G}_{F_v})$. In particular it has a real structure, and we can choose a Euclidean norm on $\operatorname{Re}(\widehat{\mathfrak{h}_v})$ invariant under the Weyl group, yielding an invariant norm $|| \cdot ||_v$ on $\hat{\mathfrak{h}}_v$. Clearly we can also take this norm to be invariant under outer automorphisms. The discrete part of the spectral side of the trace formula for **G** is *formally* the sum over $t \in \mathbb{R}_{\geq 0}$ of $I^{\mathbf{G}}_{\mathrm{disc},t}(f(g)dg)$, and following Arthur we considers these terms individually. Fix a finite set T of irreducible continuous representations of $\prod_{v \mid \infty} K_v$ and a compact open subgroup U of $\mathbf{G}(\mathbb{A}_{F,f})$. By [Art88, Lemma 4.1] the linear form $I_{\text{disc},t}^{\mathbf{G}}$ on the space of smooth compactly supported distributions which are bi-U-invariant and bi- $\left(\prod_{v\mid\infty}K_v\right)$ -finite with type in T is a linear combination of traces in irreducible unitary admissible representations $\pi = \otimes'_v \pi_v$ of $\mathbf{G}(\mathbb{A}_F)$ such that $\sum_{v|\infty} ||\operatorname{Im}(\nu_{\pi_v})||_v = t$. We stress that in general $I_{\operatorname{disc},t}^{\mathbf{G}}(f(g)dg)$ is not simply the sum of $m(\pi)tr(\pi(f(g)dg))$ over the set of discrete automorphic representations π for **G** such that $\sum_{v \mid \infty} ||\text{Im}(\nu_{\pi_v})||_v = t$, as there are additional terms indexed by proper Levi subgroups of G containing a given minimal one: see [Art13, 3.1.1]. These additional representations of $\mathbf{G}(\mathbb{A}_F)$ contributing to $I^{\mathbf{G}}_{\mathrm{disc},t}(f(g)dg)$, however, have non-regular infinitesimal character at all Archimedean places of F.

The stabilisation of the discrete part of the trace formula is the equality

$$I_{\mathrm{disc},t}^{\mathbf{G}}(f(g)dg) = \sum_{\mathfrak{e}} \iota(\mathfrak{e}) S_{\mathrm{disc},t}^{\mathbf{H}}(f'(h)dh)$$
(10)

where the sum is over the isomorphism classes of elliptic endoscopic data $\mathbf{e} = (\mathbf{H}, \mathcal{H}, s, \xi)$ for \mathbf{G} , and $\iota(\mathbf{e}) = \tau(\mathbf{G})\tau(\mathbf{H})^{-1}|\operatorname{Out}(\mathbf{e})|^{-1}$. For any such isomorphism class a representative \mathbf{e} is chosen, along with a compatible L-embedding ${}^{L}\xi : {}^{L}\mathbf{H} \to {}^{L}\mathbf{G}$. As explained before this yields canonical adélic transfer factors, and the smooth, compactly supported distribution f'(h)dh on the right hand side of (10) can be any transfer of f(g)dg. It is possible to choose $\prod_{v|\infty} f'_v(h_v)$ to be bi-finite for any given maximal compact subgroup of $\mathbf{H}(\mathbb{R} \otimes_{\mathbb{Q}} F)$. The right hand side makes sense because Arthur has shown that the linear forms $f'(h)dh \mapsto S_{\operatorname{disc},t}^{\mathbf{H}}(f'(h)dh)$, defined inductively by the equality (10) with **G** replaced by **H**, are stable. Note that there is only a finite number of isomorphism classes of elliptic endoscopic data unramified outside a finite set of places of F by [Lan83, Lemme 8.12], and combined with [Kot86, Proposition 7.5] this implies that there are only finitely many non-zero terms in the sum on the right hand side of (10).

Arthur gives a spectral expansion for the terms $S_{\text{disc},t}^{\mathbf{H}}(f'(h)dh)$, under the assumption that the stable orbital integrals of f'(h)dh are invariant under $\text{Out}(\mathbf{H}_{1,F_v}) \times \text{Out}(\mathbf{H}_{2,F_v})$ at all places v of F. By [Art13, Lemma 3.3.1, Proposition 3.4.1 and Theorem 4.1.2],

$$S_{\mathrm{disc},t}^{\mathbf{H}}(f'(h)dh) = \sum_{\substack{\psi' \in \widetilde{\Psi}(\mathbf{H})\\ \text{s.t. } t(\psi') = t}} m_{\psi'} |\mathcal{S}_{\psi'}|^{-1} \sigma(\overline{\mathcal{S}}_{\psi'}^{0}) \epsilon_{\psi'}(s_{\psi'}) \Lambda_{\psi'}(f'(h)dh).$$

Here $\Lambda_{\psi'} = \prod_{v} \Lambda_{\psi'_{v}}$ is the stable linear form associated to $\psi' = (\psi'_{1}, \psi'_{2})$ (see (1)), and $\widetilde{\Psi}(\mathbf{H}) = \widetilde{\Psi}(\mathbf{H}_{1}) \times \widetilde{\Psi}(\mathbf{H}_{2})$. In general the set $\widetilde{\Psi}(\mathbf{H}_{i})$ properly contains the set of *discrete* parameters $\widetilde{\Psi}_{\text{disc}}(\mathbf{H}_{i})$ reviewed in section 2.4. These substitutes for non-discrete Arthur-Langlands parameters are also defined as formal unordered direct sums $\psi'_{i} = \bigoplus_{j} \pi_{i,j} [d_{i,j}]^{\boxplus l_{i,j}}$, where $\pi_{i,j}$ is a unitary automorphic cuspidal representation of $\mathbf{GL}_{n_{i,j}}/F$, $d_{i,j} \geq 1$, $l_{i,j} \geq 1$, the pairs $(\pi_{i,j}, d_{i,j})$ are distinct, and if $\pi_{i,j}$ is not self-dual then there is a unique index k such that $\pi_{i,k}^{\vee} \simeq \pi_{i,j}$ and $d_{i,k} = d_{i,j}$, and it satisfies $l_{i,k} = l_{i,j}$. As in the discrete case one can consider at each place v of F the localisation $\psi'_{i,v}$, an Arthur-Langlands parameter $\mathrm{WD}_{F_{v}} \times \mathrm{SL}_{2}(\mathbb{C}) \to {}^{L}\mathbf{H}_{i}$ which is well-defined up to the action of $\mathrm{Aut}({}^{L}\mathbf{H}_{i})$. The number $t(\psi')$ is defined as the norm of the imaginary part of the infinitesimal character of ${}^{L}\xi(\psi')$. We do not recall the definition of $\sigma(\overline{S}_{\psi'}^{0})$ because eventually we will not need it.

By [Mez13, Lemma 24] or [Wal, Corollaire 2.8], the transfer correspondence $f_v(g_v)dg_v \leftrightarrow f'_v(h_v)dh_v$ at an Archimedean place v of F is compatible with infinitesimal characters in the following sense. The L-embedding ${}^L\xi$ yields an algebraic mapping from the complex algebraic variety of infinitesimal characters for \mathbf{H}_{F_v} to the analogue variety for \mathbf{G}_{F_v} . For the cases considered in this paper this is particularly simple since it is not necessary to take a z-extension of \mathbf{H} , and ${}^L\xi$ is defined between L-groups formed with Galois groups rather than Weil groups. Dually we get morphisms of \mathbb{C} -algebras $({}^L\xi)^*_v : \mathcal{Z}(\mathfrak{g}_v) \to \mathcal{Z}(\mathfrak{h}_v)$. The algebra $\mathcal{Z}(\mathfrak{g}_v)$ acts on the space of bi- K_v -finite smooth compactly supported distributions on $\mathbf{G}(F_v)$, and preserves any subspace defined by restricting the K_v -types. The aforementioned compatibly is that if $f'_v(h_v)dh_v$ is a transfer of $f_v(g_v)dg_v$. Therefore, for any infinitesimal character $\nu = (\nu_v)_{v|\infty}$ for \mathbf{G} such that $\sum_{v|\infty} ||\mathrm{Im}(\nu_v)||_v = t$, the stabilisation of the trace formula (10) can be refined as

$$I_{\mathrm{disc},\tilde{\nu}}^{\mathbf{G}}(f(g)dg) = \sum_{\mathfrak{e}} \iota(\mathfrak{e}) \sum_{\psi' \in \tilde{\Psi}(\mathbf{H},\tilde{\nu})} m_{\psi'} |\mathcal{S}_{\psi'}|^{-1} \sigma(\overline{S}_{\psi'}^{0}) \epsilon_{\psi'}(s_{\psi'}) \Lambda_{\psi'}(f'(h)dh)$$

where $\widetilde{\Psi}(\mathbf{H}, \widetilde{\nu})$ is the subset of $\widetilde{\Psi}(\mathbf{H})$ consisting of those ψ' such that for any Archimedean place v, the infinitesimal character of ${}^{L}\xi(\psi'_{v})$ belongs to the $\operatorname{Out}(\mathbf{G}_{F_{v}})$ -orbit of ν_{v} , and

 $I_{\mathrm{disc},\tilde{\nu}}^{\mathbf{G}}$ is obtained from the linear combination $I_{\mathrm{disc},t}^{\mathbf{G}}$ of traces in irreducible unitary admissible representations of $\mathbf{G}(\mathbb{A}_F)$ by keeping only those representations whose infinitesimal character belongs to $\tilde{\nu}$. Of course this compatibility was already used in the proof of the trace formula and in [Art13].

From now on we assume that ν_v is algebraic regular for all $v \in S$. We claim that for $v \in S$ the Arthur-Langlands parameter ${}^{L}\xi(\psi'_{v})$, whose $\widehat{\mathbf{G}}$ -conjugacy class is only defined up to outer conjugacy if \mathbf{G} is even orthogonal, is an Adams-Johnson parameter in the sense of section 3.2.2. We only have to check that the restriction of ψ'_v to $W_{\overline{F_v}} \times \{1\} \subset$ $W_{F_v} \times \mathrm{SL}_2(\mathbb{C})$ is bounded. For any *i* and *j* the infinitesimal character of the Langlands parameter $W_{F_v} \to \operatorname{GL}_{n_{i,j}}(\mathbb{C})$ associated to $(\pi_{i,j})_v$ or $(\pi_{i,j})_v |\det|^{1/2}$ is algebraic, and by Clozel's purity lemma [Clo88, Lemme 4.9] this implies that $(\pi_{i,j})_v$ is tempered, i.e. its Langlands parameter is bounded.

Since $S \neq \emptyset$, this implies that $L_{\xi}(\psi')$ is discrete, and in particular ψ' itself is discrete, which implies that $\sigma(\overline{S}^0_{\psi'}) = 1$. Moreover, as we recalled above, the contributions from proper Levi subgroups of \mathbf{G} to the discrete part of the spectral side of the trace formula for G have non-regular infinitesimal character at all Archimedean places of F, thus the stabilisation of the trace formula becomes the more concrete equality

$$\operatorname{tr}\left(f(g)dg \,\middle|\, \mathcal{A}^{2}(\mathbf{G},\tilde{\nu})\right) = \sum_{\mathfrak{e}} \iota(\mathfrak{e}) \sum_{\psi' \in \widetilde{\Psi}_{\mathbf{G}-\operatorname{disc}}(\mathbf{H},\tilde{\nu})} m_{\psi'} |\mathcal{S}_{\psi'}|^{-1} \epsilon_{\psi'}(s_{\psi'}) \Lambda_{\psi'}(f'(h)dh)$$

where $\widetilde{\Psi}_{\mathbf{G}-\text{disc}}(\mathbf{H},\widetilde{\nu}) = \widetilde{\Psi}(\mathbf{H},\widetilde{\nu}) \cap \widetilde{\Psi}_{\mathbf{G}-\text{disc}}(\mathbf{H})$. The right hand side can also be written as

$$\sum_{\mathfrak{e}} \iota(\mathfrak{e}) \sum_{\substack{\dot{\psi}' \in \Psi_{\mathbf{G}-\operatorname{disc}}(\mathbf{H},\tilde{\nu})\\ \text{up to } \widehat{\mathbf{H}}-\operatorname{conj.}}} |\mathcal{S}_{\psi'}|^{-1} \epsilon_{\psi'}(s_{\psi'}) \Lambda_{\psi'}(f'(h)dh).$$

We now want to express this using discrete parameters for \mathbf{G} , using Proposition 2.4.1. Let us write $\dot{\psi} = {}^L \xi \circ \dot{\psi}'$, and recall that s defines an element of $\mathcal{S}_{\dot{\psi}}$. By [Art13, Lemma 4.4.1] we have $\epsilon_{\psi'}(s_{\psi'}) = \epsilon_{\psi}(ss_{\psi})$, and by equation (8) we have

$$\Lambda_{\psi'}(f'(h)dh) = \sum_{\pi \in \Pi_{\psi}(\mathbf{G})} \langle ss_{\psi}, \pi \rangle \operatorname{tr}(\pi(f(g)dg))$$

The stabiliser of the $\widehat{\mathbf{H}}$ -conjugacy class of $\dot{\psi}'$ under the action of $\operatorname{Out}(\mathfrak{e})$ is naturally identified with

$$\left(\operatorname{Aut}(\mathfrak{e})\cap C_{\dot{\psi}}\right)/\xi(C_{\dot{\psi}'})$$

and $\left(\operatorname{Aut}(\mathfrak{e}) \cap C_{\dot{\psi}}\right) / Z(\widehat{\mathbf{G}})^{\operatorname{Gal}_F} = \operatorname{Cent}(s, \mathcal{S}_{\dot{\psi}})$ is simply equal to $\mathcal{S}_{\dot{\psi}}$ because $\mathcal{S}_{\dot{\psi}}$ is abelian. Thus there are $\underbrace{|\operatorname{Out}(\mathfrak{e})| \times |\mathcal{S}_{\dot{\psi}'}| \times |\xi(Z(\widehat{\mathbf{H}})^{\operatorname{Gal}_F}) / Z(\widehat{\mathbf{G}})^{\operatorname{Gal}_F}|}_{(11)}$

$$\frac{|\operatorname{Out}(\mathbf{\mathfrak{e}})| \times |\mathcal{S}_{\dot{\psi}'}| \times |\xi(Z(\mathbf{H})^{\operatorname{Gal}_F})/Z(\mathbf{G})^{\operatorname{Gal}_F}|}{|\mathcal{S}_{\dot{\psi}}|}$$
(11)

 $\widehat{\mathbf{H}}$ -conjugacy classes in $\Psi_{\mathbf{G}-\text{disc}}(\mathbf{H})$ that map to the $\widehat{\mathbf{G}}$ -conjugacy class of $(\dot{\psi}, sZ(\widehat{\mathbf{G}}))$. The groups **G** and **H** satisfy the Hasse principle, therefore $\tau(\mathbf{G}) = |\pi_0(Z(\widehat{\mathbf{G}})^{\operatorname{Gal}_F})|$ and $\tau(\mathbf{H}) = |\pi_0(Z(\widehat{\mathbf{H}})^{\operatorname{Gal}_F})|, \text{ and finally we can conclude that (11) equals } |\mathcal{S}_{\dot{\psi}'}|\iota(\mathfrak{e})^{-1}|\mathcal{S}_{\dot{\psi}}|^{-1}.$ By grouping the terms corresponding to $\operatorname{Out}(\mathfrak{e})$ -orbits in the sum, we get

$$\operatorname{tr}\left(f(g)dg \,\middle|\, \mathcal{A}^{2}(\mathbf{G},\tilde{\nu})\right) = \sum_{\substack{\dot{\psi}\in\Psi_{\operatorname{disc}}(\mathbf{G},\tilde{\nu})\\ \operatorname{up to}\, \hat{\mathbf{G}}-\operatorname{conj}}} \sum_{\substack{s\in\mathcal{S}_{\psi}\\ \pi\in\Pi_{\psi}(\mathbf{G})}} \sum_{\substack{\pi\in\Pi_{\psi}(\mathbf{G})\\ \psi\in\Psi_{\operatorname{disc}}(\mathbf{G},\tilde{\nu})\\ \operatorname{up to}\, \hat{\mathbf{G}}-\operatorname{conj}}} \sum_{\substack{\pi\in\Pi_{\psi}(\mathbf{G})\\ \operatorname{s.t.}\langle\cdot,\pi\rangle=\epsilon_{\psi}}} \operatorname{tr}(\pi(f(g)dg))$$

- Remark 4.0.2. 1. We formulated Theorem 4.0.1 using traces of distributions on the discrete automorphic spectrum, but of course this is equivalent to a formulation similar to that of Theorem 1.5.2 in [Art13].
 - 2. If **G** is such that $\mathbf{G}(\mathbb{R} \otimes_{\mathbb{Q}} F)$ is compact, in which case one may take for S the set of all real places of F, Theorem 4.0.1 describes the full automorphic spectrum of **G**.

References

- [AJ87] Jeffrey Adams and Joseph F. Johnson, Endoscopic groups and packets of nontempered representations, Compositio Math. 64 (1987), no. 3, 271–309.
- [AMR] Nicolás Arancibia, Colette Moeglin, and David Renard, Paquets d'arthur des groupes classiques et unitaires, http://arxiv.org/abs/1507.01432.
- [Art88] James Arthur, The invariant trace formula. II. Global theory, J. Amer. Math. Soc. 1 (1988), no. 3, 501–554.
- [Art89] _____, Unipotent automorphic representations: conjectures, Astérisque (1989), no. 171-172, 13-71, Orbites unipotentes et représentations, II.
- [Art01] _____, A stable trace formula. II. Global descent, Invent. Math. 143 (2001), no. 1, 157–220.
- [Art02] _____, A stable trace formula. I. General expansions, J. Inst. Math. Jussieu 1 (2002), no. 2, 175–277.
- [Art03] _____, A stable trace formula. III. Proof of the main theorems, Ann. of Math.
 (2) 158 (2003), no. 3, 769–873.
- [Art13] James Arthur, The Endoscopic Classification of Representations: Orthogonal and Symplectic groups, American Mathematical Society Colloquium Publications, vol. 61, American Mathematical Society, 2013.
- [BG10] Kevin Buzzard and Toby Gee, The conjectural connections between automorphic representations and Galois representations, http://arxiv.org/abs/1009. 0785v1, 2010.
- [BJ79] A. Borel and H. Jacquet, Automorphic forms and automorphic representations, Automorphic forms, representations and L-functions (Proc. Sympos. Pure Math., Oregon State Univ., Corvallis, Ore., 1977), Part 1, Proc. Sympos. Pure Math., XXXIII, Amer. Math. Soc., Providence, R.I., 1979, pp. 189–207.
- [Bor79] A. Borel, Automorphic L-functions, Automorphic forms, representations and Lfunctions (Proc. Sympos. Pure Math., Oregon State Univ., Corvallis, Ore., 1977), Part 2, Proc. Sympos. Pure Math., XXXIII, Amer. Math. Soc., Providence, R.I., 1979, pp. 27–61.
- [Cha09] Ngo Bao Chau, Report on the proof of some conjectures on orbital integrals in Langlands' program, Vietnam J. Math. 37 (2009), no. 2-3, 127–140.
- [Che04] Gaëtan Chenevier, Familles p-adiques de formes automorphes pour GL_n , J. Reine Angew. Math. **570** (2004), 143–217.

- [CHL11] Raf Cluckers, Thomas Hales, and François Loeser, Transfer principle for the fundamental lemma, On the stabilization of the trace formula, Stab. Trace Formula Shimura Var. Arith. Appl., vol. 1, Int. Press, Somerville, MA, 2011, pp. 309–347.
- [CL] Gaëtan Chenevier and Jean Lannes, Formes automorphes et voisins de kneser des réseaux de niemeier, http://arxiv.org/abs/1409.7616.
- [CLH] Ana Caraiani and Bao Le Hung, On the image of complex conjugation in certain galois representations, http://arxiv.org/abs/1409.2158.
- [Clo88] Laurent Clozel, Motifs et Formes Automorphes: Applications du Principe de Fonctorialité, Automorphic Forms, Shimura Varieties, and L-functions (Laurent Clozel and James S. Milne, eds.), vol. 1, 1988, Proceedings of a Conference held at the University of Michigan, Ann Arbor.
- [CR14] Gaëtan Chenevier and David Renard, Level one algebraic cuspforms of classical groups of small ranks, Mem. Amer. Math. Soc. 9999 (2014), no. 9999, 120.
- [CS80] William Casselman and J. Shalika, The unramified principal series of p-adic groups. II: The Whittaker function., Compos. Math. 41 (1980), 207–231.
- [Hal95] Thomas C. Hales, On the fundamental lemma for standard endoscopy: reduction to unit elements, Canad. J. Math. 47 (1995), no. 5, 974–994.
- [Joh84] Joseph F. Johnson, Lie algebra cohomology and the resolution of certain Harish-Chandra modules, Math. Ann. 267 (1984), no. 3, 377–393.
- [Kala] Tasho Kaletha, Global rigid inner forms and multiplicities of discrete automorphic representations, available at https://web.math.princeton.edu/ ~tkaletha/.
- [Kalb] _____, Rigid inner forms of real and p-adic groups, available at https://web. math.princeton.edu/~tkaletha/.
- [Kalc] Tasho Kaletha, Rigid inner forms vs isocrystals, http://arxiv.org/abs/1502. 00650.
- [KMSW] Tasho Kaletha, Alberto Minguez, Sug Woo Shin, and Paul-James White, Endoscopic classification of representations: Inner forms of unitary groups, http: //arxiv.org/abs/1409.3731.
- [Kot83] Robert E. Kottwitz, Sign changes in harmonic analysis on reductive groups, Trans. Amer. Math. Soc. 278 (1983), no. 1, 289–297.
- [Kot84] _____, Stable trace formula: cuspidal tempered terms, Duke Math. J. 51 (1984), no. 3, 611-650.
- [Kot86] Robert Kottwitz, Stable trace formula: elliptic singular terms, Mathematische Annalen 275 (1986), no. 3, 365–399.

- [Kot90] Robert E. Kottwitz, Shimura varieties and λ-adic representations, Automorphic forms, Shimura varieties, and L-functions, Vol. I (Ann Arbor, MI, 1988), Perspect. Math., vol. 10, Academic Press, Boston, MA, 1990, pp. 161–209.
- [KS] Robert E. Kottwitz and Diana Shelstad, On splitting invariants and sign conventions in endoscopic transfer, http://arxiv.org/abs/1201.5658.
- [KS99] _____, Foundations of twisted endoscopy, Astérisque (1999), no. 255, vi+190.
- [Lan83] R. P. Langlands, Les débuts d'une formule des traces stable, Publications Mathématiques de l'Université Paris VII, vol. 13, Université de Paris VII, U.E.R. de Mathématiques, Paris, 1983.
- [Lan89] _____, On the classification of irreducible representations of real algebraic groups, Representation theory and harmonic analysis on semisimple Lie groups, Math. Surveys Monogr., vol. 31, Amer. Math. Soc., Providence, RI, 1989, pp. 101-170.
- [LL79] J.-P. Labesse and R. P. Langlands, *L-indistinguishability for* SL(2), Canad. J. Math. **31** (1979), no. 4, 726–785.
- [Loe11] David Loeffler, Overconvergent algebraic automorphic forms, Proc. London Math. Soc. 102 (2011), no. 2, 193–228.
- [LS90] R. Langlands and D. Shelstad, Descent for transfer factors, The Grothendieck Festschrift, Vol. II, Progr. Math., vol. 87, Birkhäuser Boston, Boston, MA, 1990, pp. 485–563.
- [Mez13] Paul Mezo, Character identities in the twisted endoscopy of real reductive groups, Mem. Amer. Math. Soc. 222 (2013), no. 1042, vi+94.
- [Mok15] Chung Pang Mok, Mem. Amer. Math. Soc. 235 (2015), no. 1108, vi+248.
- [Ngô10] Bao Châu Ngô, Le lemme fondamental pour les algèbres de Lie, Publ. Math. Inst. Hautes Études Sci. (2010), no. 111, 1–169.
- [PR94] Vladimir Platonov and Andrei Rapinchuk, Algebraic groups and number theory, Pure and Applied Mathematics, vol. 139, 1994.
- [Ser94] Jean-Pierre Serre, Cohomologie galoisienne, fifth ed., Lecture Notes in Mathematics, vol. 5, Springer-Verlag, Berlin, 1994.
- [She08a] D. Shelstad, Tempered endoscopy for real groups. I. Geometric transfer with canonical factors, Representation theory of real reductive Lie groups, Contemp. Math., vol. 472, Amer. Math. Soc., Providence, RI, 2008, pp. 215–246.
- [She08b] _____, Tempered endoscopy for real groups. III. Inversion of transfer and Lpacket structure, Represent. Theory 12 (2008), 369–402.

- [She10] _____, Tempered endoscopy for real groups. II. Spectral transfer factors, Automorphic forms and the Langlands program, Adv. Lect. Math. (ALM), vol. 9, Int. Press, Somerville, MA, 2010, pp. 236–276.
- [Taïa] Olivier Taïbi, Dimensions of spaces of level one automorphic forms for split classical groups using the trace formula, http://arxiv.org/abs/1406.4247.
- [Taïb] _____, Eigenvarieties for classical groups and complex conjugations in galois representations, http://arxiv.org/abs/1203.0225.
- [Wal] Jean-Loup Waldspurger, Stabilisation de la formule des traces tordue I: endoscopie tordue sur un corps local, http://arxiv.org/abs/1401.4569.
- [Wal95] J.-L. Waldspurger, Une formule des traces locale pour les algèbres de Lie padiques, J. Reine Angew. Math. 465 (1995), 41–99.
- [Wal97] _____, Le lemme fondamental implique le transfert, Compositio Math. 105 (1997), no. 2, 153–236.
- [Wal06] _____, Endoscopie et changement de caractéristique, J. Inst. Math. Jussieu 5 (2006), no. 3, 423–525.
- [Wal10] Jean-Loup Waldspurger, Les facteurs de transfert pour les groupes classiques: un formulaire, Manuscripta Math. 133 (2010), no. 1-2, 41–82.